

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Attanasio, Orazio; Cattan, Sarah; Fitzsimons, Emla; Meghir, Costas; Rubio-Codina, Marta

# Working Paper Estimating the production function for human capital: Results from a randomized control trial in Colombia

Economic Growth Center Discussion Paper, No. 1046

**Provided in Cooperation with:** Yale University, Economic Growth Center (EGC)

*Suggested Citation:* Attanasio, Orazio; Cattan, Sarah; Fitzsimons, Emla; Meghir, Costas; Rubio-Codina, Marta (2015) : Estimating the production function for human capital: Results from a randomized control trial in Colombia, Economic Growth Center Discussion Paper, No. 1046, Yale University, Economic Growth Center, New Haven, CT

This Version is available at: https://hdl.handle.net/10419/109690

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



# WWW.ECONSTOR.EU

# ECONOMIC GROWTH CENTER YALE UNIVERSITY

P.O. Box 208269 New Haven, CT 06520-8269

http://www.econ.yale.edu/~egcenter



Economic Growth Center Discussion Paper No. 1046 Cowles Foundation Discussion Paper No. 1987

# **Estimating the Production Function for Human Capital: Results from a Randomized Control Trial in Colombia**

**Orazio Attanasio** University College London

Sarah Cattan Institute for Fiscal Studies

**Emla Fitzsimons** UCL Institute of Education and Institute for Fiscal Studies

**Costas Meghir** Yale University NBER, IZA, and Institute for Fiscal Studies

# Marta Rubio-Codina

Institute for Fiscal Studies and Inter-American Development Bank

Notes: Center discussion papers are preliminary materials circulated to stimulate discussion and critical comments.

# Estimating the Production Function for Human Capital: Results from a Randomized Control Trial in Colombia

Orazio Attanasio, Sarah Cattan, Emla Fitzsimons, Costas Meghir, and Marta Rubio-Codina<sup>\*</sup>

February 11, 2015

#### Abstract

We examine the channels through which a randomized early childhood intervention in Colombia led to significant gains in cognitive and socioemotional skills among a sample of disadvantaged children. We estimate production functions for cognitive and socio-emotional skills as a function of maternal skills and child's past skills, as well as material and time investments that are treated as endogenous. The effects of the program can be fully explained by increases in parental investments, which have strong effects on outcomes and are complementary to both maternal skills and child's past skills.

<sup>\*</sup>Attanasio: University College London and Institute for Fiscal Studies (o.attanasio@ucl.ac.uk). Cattan: Institute for Fiscal Studies (sarah\_c@ifs.org.uk). Fitzsimons: UCL Institute of Education and Institute for Fiscal Studies (e.fitzsimons@ioe.ac.uk). Meghir: Yale University, NBER, IZA and Institute for Fiscal Studies (c.meghir@yale.edu). Rubio-Codina: Institute for Fiscal Studies and Inter-American Development Bank (marta\_r@ifs.org.uk). We thank participants at the NBER Summer Institute, Barcelona GSE Summer Forum and Montreal CIREQ Applied Microeconomics on Fertility and Child Development and seminars at Stanford University, University of Chicago, Oxford University, Cornell University, Bristol University and the Institute for Fiscal Studies for their comments. We are grateful to the Economic and Social Research Council (Grant ES/G015953/1), the Inter-American Development Bank, the International Growth Centre, and the World Bank for funding the intervention and data collection. Some of this research was financed by the European Research Council's Advanced Grant 249612 and by the Grand Challenges Canada Prime Award 0072-03 (sub-award reference number 560450). Sarah Cattan gratefully acknowledges financial assistance from the British Academy Postdoctoral Fellowship pf140104, as well as from the European Research Council's Grant Agreement No. 240910. Costas Meghir thanks the Cowles foundation and the ISPS at Yale for financial assistance. All errors are the responsibility of the authors.

# 1 Introduction

The first five years of life lay the basis for lifelong outcomes (Almond and Currie, 2011). Due to rapid brain development and its malleability during the early years (Knudsen, 2004; Knudsen et al., 2006), investments during this period may play a crucial role in the process of human capital accumulation. At this time, many children are, however, exposed to risk factors such as poverty, malnutrition and non-stimulating home environments preventing them from reaching their full potential and perpetuating poverty, particularly in developing countries (McGregor et al., 2007).

There is increasing evidence that early childhood interventions can alleviate the consequences of these detrimental factors in a long-lasting fashion. Examples include the Jamaica study (Gertler et al., 2014; Grantham-McGregor et al., 1991; Walker et al., 2011) and the Perry Preschool program (Heckman et al., 2010). However, less is known about the behavioural mechanisms through which these interventions affect children and their families.

This paper aims to examine the channels through which an early years intervention in Colombia led to significant gains in cognitive and socio-emotional skills among a sample of disadvantaged children. The intervention we study was a randomized control trial that targeted children between 12 and 24 months old, for a period of 18 months, in families who are beneficiaries of the conditional cash transfer program in Colombia (*Familias en Acción*). Its structure mirrored that of the Jamaica intervention in that it included a psychosocial stimulation component and a micro-nutrient supplementation component. The psychosocial stimulation program, which we focus on in this paper, provided weekly home visits to mothers of children, for a period of 18 months, with the aim of improving parenting practices in the early years and beyond.

The short-term impact evaluation of the intervention reveals that psychosocial stimulation had significant positive effects on the language and cognitive development of children who received the home visits (Attanasio et al., 2014). In what follows, we reproduce these results and also show impacts of the intervention in other dimensions, such as indicators of socio-emotional development. However, these results could have been generated by a number of different mechanisms. In addition to the weekly, one-hour home visit during which the child and their mother interacted with the home visitor, the intervention could have altered parental investment behavior by making them aware of the importance of early investments and informing them about parenting practices that enhance the child's learning at home.

In order to shed light on the mechanisms through which the stimulation program affected child development, we estimate parents' investment functions and the production functions for cognitive and socio-emotional skills. Following Cunha, Heckman, and Schennach (2010), we model the accumulation of future skills as a process that is determined by the child's current stock of skills, parents' investments and parental human capital as well as (unobservable) shocks. This technology is non-linear and allows the degree of substitutability between inputs to be determined from the data. We consider two types of investment (time and commodities) and allow parental choices to be endogenously determined by estimating investment functions that depend on resources and prices. This approach provides a natural framework to interpret and understand the potential channels through which the psychosocial stimulation component of the intervention could have boosted the skills of treated children. In particular, the intervention could have shifted the distribution of parental investments and/or changed the parameters of the production function, for example by making parents more productive or effective.

To estimate the production functions for cognitive and socio-emotional skills, we use data we collected both before and after the intervention in Colombia. The data contain very rich measures of child development, maternal skills and parental investments. Even with such rich data, estimating the parameters governing the skill formation process remains challenging for two reasons. First, inputs and outputs are likely to be measured with error. Second, inputs, especially investments, can be endogenous if parental investment decisions respond to shocks or inputs that are unobserved to the econometrician. To deal with the measurement error issue, we use dynamic latent factor models as Cunha, Heckman, and Schennach (2010). The endogeneity of investment is taken into account by implementing a control function approach, as in Attanasio, Meghir, and Nix (2015), whose estimation procedure we adopt here. The exclusion restrictions needed for identification are justified by the economics of the model.

Our estimates of the production function reveal a series of interesting and important findings. First, in line with the existing literature, we find strong evidence that a child's current stock of skills fosters the development of future skills.<sup>1</sup> Second, and also in line with the existing literature, we find that current skills, parental investments and maternal human capital are complementary in the production of future skills. Parental investments matter greatly for the accumulation of skills. In particular, material investments seem to matter more for cognitive skills, while time investments seem to matter more for socio-emotional skills. Our results indicate that the parameters that determine the productivity of investment greatly depend on whether investment is considered endogenous. Ignoring the fact that parents choose investment leads to a large downward bias of the estimated productivity of investment in the production functions, therefore indicating that parents use investment to compensate for negative shocks. Interestingly, this result is also obtained by Cunha, Heckman, and Schennach (2010) and Attanasio, Meghir, and Nix (2015), yet in very different contexts.<sup>2</sup>

With respect to the mechanisms through which the intervention operated, we find that the intervention significantly increased parental investments among treated families compared to non-treated ones. At the same time, there are no significant shifts in the parameters of the production function induced by the intervention. These two findings mean that the gains in cognitive and socio-emotional skills among children who received the intervention are fully

<sup>&</sup>lt;sup>1</sup>These features of the technology of skill formation are often referred to as *self-productivity* and *cross-productivity* (Cunha et al., 2006).

<sup>&</sup>lt;sup>2</sup>The former use the Children of the National Longitudinal Survey of Youth 1979, a longitudinal panel following the children of a representative sample of women born between 1956 and 1964 in the US. The latter use the Young Lives Survey for India, a longitudinal survey following the lives of children in two age-groups: a Younger Cohort of 2,000 children who were aged between 6 and 18 months when Round 1 of the survey was carried out in 2002, and an Older Cohort of 1,000 children then aged between 7.5 and 8.5 years. The survey was carried out again in late 2006 and in 2009 (when the younger children were about 8 the same age as the Older Cohort when the research started in 2002).

explained by the shift in investments.

Our findings make important contributions to the literature on human capital development, especially during the early years. To the best of our knowledge, our paper is the first to estimate the technology of skill formation in the first three years of life and to quantify the size of the dynamic complementarities between different domains of human development at such young ages. Characterising the production function at various ages is key for the identification of critical periods that are important to target for the development of particular skills. Our paper and that of Attanasio, Meghir, and Nix (2015) are the first to estimate non-linear production functions in a developing country context.<sup>3</sup> Helmers and Patnam (2011) estimate production functions with Indian data, but they rely on a linear technology, which implies that inputs are perfect substitutes for each other. Our results show that this assumption is strongly rejected by the data and that accounting for complementarities between inputs is of high importance. In this regard, our results are strikingly consistent with those of Cunha, Heckman, and Schennach (2010) and Attanasio, Meghir, and Nix (2015), neither of whom can reject that the technology of skill formation is Cobb-Douglas. Finally, we are the first to account for multiple types of investments in children. We establish that distinguishing between material and time investments is important for our understanding of skill formation in the early years. These findings have important implications for the design of future interventions.

While there is a vast literature evaluating the impact of early childhood

<sup>&</sup>lt;sup>3</sup>Attanasio, Meghir, and Nix (2015) estimate nonlinear production functions for cognition and health in India for children from 5-15, using the Young Lives Survey.

interventions on child development, our paper innovates by complementing the information obtained from a randomized controlled trial of a specific intervention with a completely specified model of skill formation and parental investment in order to understand the mechanisms behind the observed impacts. In this sense, our paper shares the motivation of Heckman, Pinto, and Savelyev (2013), who are interested in the channels through which the Perry Pre-School Program produced gains in adult outcomes. Our focus and methodology, however, are different: Heckman, Pinto, and Savelyev (2013) perform a mediation analysis that decomposes linearly the treatment effects on adult outcomes into components attributable to early changes in different personality traits. We use a structural model in which parents make investment choices and human capital accumulates according to a completely specified production function to interpret and explain the impacts induced by a successful intervention. We explicitly test alternative and specific hypothesis about the origin of the impacts. Despite these differences, along with Heckman, Pinto, and Savelyev (2013) and a few other papers (Attanasio, Meghir, and Santiago, 2012; Duflo, Hanna, and Ryan, 2012; Todd and Wolpin, 2006), our paper illustrates how data from randomized trials can be profitably combined with behavioral models to go beyond the estimation of experimentally induced treatment effects and interpret the mechanisms underlying them, a crucial step for policy analysis.

The paper proceeds as follows. Section 2 describes the intervention and the data collected pre- and post-intervention and summarizes the short-term impacts of the intervention. Section 3 presents the theoretical framework we use and discusses its identification. Section 4 describes our estimation strategy. Section 5 presents the estimates of the model and discusses their implications for our understanding of the intervention. Section 6 concludes.

# 2 The intervention, its evaluation and its impacts

Although some influential studies have shown that well-designed and welltargeted interventions can achieve spectacular results that are sustained over long periods of time, a key challenge remains in the design of interventions that can be deployed on a large scale at reasonable cost whilst at the same time maintaining the quality that underlies the observed impacts. In this study, we use data from the evaluation of an intervention that was designed as an effectiveness rather than an efficacy trial as it was deployed on a relatively large scale and was delivered by local people. In this section, we give some details on the intervention and its evaluation design.

# 2.1 The intervention design

The integrated early childhood program analyzed in this paper was targeted at children aged between 12 and 24 months living in families receiving the Colombian CCT program (*Familias en Acción*), which targets the poorest 20% of households in the country. The intervention lasted 18 months, starting in early 2010. Appendix A contains a detailed description of the program's design, implementation and delivery. Here we summarize the key aspects.

The program was implemented in semi-urban municipalities in three regions of central Colombia, covering an area three times the size of England. It had two components: psychological stimulation and micronutrient supplementation. In this paper, we focus on the psychosocial stimulation arm of the program. The stimulation curriculum was based on the Jamaican home visiting model, which obtained positive short- and long-term effects (Gertler et al., 2014; Grantham-McGregor et al., 1991; Walker et al., 2006). The protocols designed by Grantham-McGregor et al. (1991) for Jamaica were adapted to be culturally appropriate for Colombia. The aims of the home visits were to improve the quality of maternal-child interactions and to assist mothers to participate in developmentally-appropriate learning activities, centered around daily routines and using household resources.

Two key innovations vis-a-vis the Jamaican intervention were made so as to incorporate scalability and sustainability. The first was that the intervention was implemented on a much larger scale than in Jamaica. The second was that home visitors were drawn from a network of local women, generated by the administrative set-up of the CCT program. *Familias en Acción*, has, as many other CCTs, an important social function. Every 50-60 beneficiaries elect a representative who is in charge of organising social activities and who acts as mediators between them and the program administrators. These women, known as *Madre Líderes* (MLs), are beneficiaries of the program themselves and are typically more entrepreneurial and proactive than the average beneficiary. These characteristics marked them out as potentially effective home visitors.

The intervention was evaluated through a clustered randomized control trial. Within each of the three large regions, 32 municipalities were randomly assigned to one of 4 groups: (i) psychosocial stimulation, (ii) micronutrient supplementation, (iii) both, and (iv) control. Assignment to treatment was via cluster-level randomization. In each municipality, 3 MLs were selected and the children of the beneficiary households represented by each of these MLs and aged 12-24 months, were recruited to the study. Therefore, there was a total of 1,429 children living in 96 towns in central Colombia.

We conducted a baseline survey before the intervention started and a follow-up survey when it ended 18 months later. The surveys took place in the household, and children's development was measured directly by psychologists in community centers. The household surveys contain information on a rich set of socio-economic and demographic characteristics as well as less standard variables such as children's food intake, pre-school participation, maternal verbal ability and mental health, and maternal knowledge and information, amongst other things. We also collected information on stimulation in the home as reported by the mother, using the UNICEF Family Care Indicators (FCI) (Frongillo, Sywulka, and Kariger, 2003). This instrument includes questions about the types and numbers of play materials around the house and about the types and frequency of play activities the child engages in with an adult aged 15 or more (most often the mother).

Children's cognitive, language and motor development were assessed using the Bayley Scales of Infant and Toddler Development III, administered directly in community centers (Bayley, 2006). Children's language development was also assessed through maternal report using a Spanish adaptation of the short version of the MacArthur-Bates Communicative Development Inventory (Jackson-Maldonado, Marchman, and Fernald, 2012). Children's socioemotional development was also measured through maternal report using the Bates' Infant Characteristics Questionnaire (Bates, Freeland, and Lounsbury, 1979) and the Early Children's Behavior Questionnaire (Putnam, Gartstein, and Rothbart, 2006). All of these tests were administered both pre- and postintervention (using age-appropriate items), with the exception of the Early Children's Behavior Questionnaire which was only administered at follow-up. We describe them at length in Appendix B.

## 2.2 The short-term impacts of the intervention

### 2.2.1 Impacts on child development

The top panel of Table 1 summarizes the short-term impact of the intervention on measures of cognitive and socio-emotional development, some of which are reported in Attanasio et al. (2014).<sup>4</sup> The short-term impact evaluation of the home visits showed an increase of 0.24 of a standard deviation (SD) in cognitive development and an increase of 0.17 SD in receptive language, assessed using the Bayley Scales of Infant and Toddler Development (Bayley-III).<sup>5</sup>

The lower panel of the table also shows that the intervention led to an im-

<sup>&</sup>lt;sup>4</sup>At baseline, we administered the Bayley-III to 1,420 children and the survey to 1,429 households (Figure 1). We excluded from analyses two children who scored less than three standard deviations below the mean on the Bayley-III cognitive subscale. The attrition rate between baseline and follow-up for the Bayley-III sample was approximately 10.62% (n=153) across treatment arms: 36 (10.00%) of the children from the stimulation arm were not measured at follow-up, 47 (13.06%) from the supplementation arm, 39 (10.83%) from the combined arm and 31 (8.61%) from the control arm. The difference in loss among the groups was not statistically significant.

<sup>&</sup>lt;sup>5</sup>These treatment effects are slightly different from those reported in Attanasio et al. (2014) because in this paper we estimate the impact of psychosocial stimulation by pooling the two groups that received it and the two groups that did not, while Attanasio et al. (2014) estimates the impact of each of the four arms of the intervention separately.

provement in some dimensions of socio-emotional development. In particular, it resulted in a 0.07 SD decrease in the dimension of the Bates scale measuring difficult behavior; none of the other three components of the Bates scale were significantly affected, however.

As discussed in greater length in Attanasio et al. (2014), no significant impact of micro-nutrient supplementation on any child developmental outcomes were found. As a result, in this paper, we focus on understanding the effect of the psychosocial stimulation program on cognitive, language and socio-emotional development.

# 2.2.2 Suggestive evidence of mechanisms

There are various mechanisms through which the psychosocial stimulation program could have been effective in improving children's cognitive, language and socio-emotional development. The one-hour weekly visit aimed at providing mothers with information on early childhood development and demonstrating to them various developmental play activities they could repeat with their child in between weekly home visits. The materials and toys used in the visit were left in the home for the week following the visit in order to promote increased interaction (both quality and quantity) between mother and child on an ongoing basis. This should have subsequently affected positively various aspects of the child's home environment, as well as the mother, whose self-esteem, mental health<sup>6</sup> and parenting activities might have improved.

<sup>&</sup>lt;sup>6</sup>We tested for impacts of the intervention on the mother's mental health, years of education, IQ, vocabulary and maternal knowledge as measured by the Knowledge of Infant Development Inventory (KIDI) (see Appendix B for a detailed description of the scales we use to measure these dimensions). We did not detect any significant impacts on these dimensions

The lower panel of Table 1 summarizes some of the results from the shortterm evaluation. These show large increases in intervention areas in the varieties of play materials and play activities in the home, as measured by the FCI. This is indicative that one mechanism through which home visits might have improved child development was by promoting parental investments in children. In order to test this hypothesis and assess the extent to which changes in parental investments contributed to the observed impacts of the intervention, we need a framework to understand the process of child development. We use a production function to model the relationship between inputs and the output of skill, which we describe below.

# 3 The accumulation of human capital in the early years: a theoretical framework

In the previous section, we reported some of the impacts that an early years intervention had both on children developmental outcomes and on parental behavior. These estimates were straightforward to obtain due to the presence of a cluster randomized control trial designed to evaluate the intervention. We now build a theoretical framework that can be used to interpret and understand these results.

In particular, we use a production function to describe the process through which the skills of children evolve between the beginning and the end of the intervention. We refer to the baseline period as t, when children were aged

of mothers' human capital and therefore only report impacts on parental investments as measured by the FCI in Table 1. We return to this issue later in the paper.

Instrument:		Ba	yley		MacArth	ur-Bates
Item:	Cognitive	Language receptive	Language expressive	Fine motor	Vocabulary	Complex sentences
Treatment effect	$0.244^{**}$ ( $0.0621$ )	0.175** (0.0647)	0.0320 (0.0623)	0.0713 (0.0617)	0.0947 (0.0652)	0.0606 (0.0563)
Observations	1,264	1,264	1,262	1,261	1,321	1,321
Instrument		Ba	ites		Family Care In	dicators (FCI)
Item:	Unsociable	Difficult	Unadaptable	Unstoppable	Varieties of play materials	Varieties of play activities
Treatment	-0.0433 (0.0549)	-0.0758+ (0.0455)	0.0597 (0.0615)	-0.0313 (0.0535)	0.213** (0.0637)	$0.273^{**}$ (0.0499)
Observations	1,326	1,326	1,326	1,326	1,326	1,326

Table 1: Short-term impacts of psycho-social stimulation on cognition, language, and find motor development;

variable measured at follow-up on the intervention variable (a treatment dummy for psychosocial stimulation, combining children receiving stimulation alone and children receiving both stimulation and micro-nutrient supplementation) controlling for: child's sex; baseline level of the outcome (except for MacArthur-Bates "Complex sentences", where we control for baseline number of words spoken because the item measuring "Complex sentences" was not measured at baseline); and tester dummies. Standard errors are adjusted for clustering at the municipality level. \*\*, \* and + indicate significance at 1, 5, and 10%. All scores have been internally standardized non-parametrically for age and are therefore expressed in standard deviations (see Appendix B for details about the measures and the standardization procedure). between 12 to 24 months old, and to the post-intervention period as t + 1, when children were aged between 30 to 42 months old. Children's skills at time t + 1 are assumed to be a function of the vector of skills at t, of parental skills, of parental investments and of some shocks. Our first aim is to characterize such a function and estimate its parameters. We assume that parents choose investments in human capital, reflecting their taste, their resources and information about the current evolution of skills. Together with the production function we estimate an investment function. Finally, we explicitly allow for measurement error of all the relevant variables that enter the production function and are determined by it: child and parents' skills as well as parents' investment.

Within this framework, the intervention can affect the accumulation of skills through different channels. For example, the intervention can change the parameters of the production function or can change parents' investment behavior. To allow for these effects, we let some of the parameters of the production function and of the investment function depend on the (randomly allocated) intervention.

Because we only focus on the effect of the psychosocial stimulation program, we define the non-treated group (d = 0) as the group of children who did not receive the home visits (therefore including both the control group and the group who only received the micro-nutrients) and the treated group (d = 1) as the group of children who received the home visits (therefore including those who received only the home visits and those who received both the home visits and the micro-nutrients).

## 3.1 The production function for human capital

We consider a two-dimensional vector of skills, which includes cognitive and socio-emotional skills. In the baseline period, child *i*'s skills are denoted  $\theta_{i,t} = (\theta_{i,t}^C, \theta_{i,t}^S)$ , where  $\theta_{i,t}^C$  and  $\theta_{i,t}^S$  are cognitive and socio-emotional skills at *t*, respectively. At the end of the intervention, the child's skills are denoted  $\theta_{i,t+1} = (\theta_{i,t+1}^C, \theta_{i,t+1}^S)$ .

Following Cunha, Heckman, and Schennach (2010), we assume that the stock of skills in period t + 1 is determined by the baseline stock of the child's cognitive and socio-emotional skills  $\theta_{i,t}$ , the mother's cognitive and socio-emotional skills, denoted by  $P_{i,t}^C$  and  $P_{i,t}^S$  respectively, and the investments  $I_{i,t}$  made by the parents between t and t + 1.<sup>7</sup> We also allow for the effect of a variable  $\eta_{i,t}^k$  that reflects unobserved shock or omitted inputs. As with skills, parental investments  $I_{i,t}$  can be a multi-dimensional vector. Here, we distinguish between material and time investments, which we denote as  $I_{i,t}^M$  and  $I_{i,t}^T$  respectively.

For each skill, we assume the production function is of the Constant Elasticity of Substitution (CES) type, so we can write the technology of formation for skill k as follows:

$$\theta_{i,t+1}^{k} = A_{d}^{k} [\gamma_{1,d}^{k} \theta_{i,t}^{C\,\rho_{k}} + \gamma_{2,d}^{k} \theta_{i,t}^{S\,\rho_{k}} + \gamma_{3,d}^{k} P_{i,t}^{C\,\rho_{k}} + \gamma_{4,d}^{k} P_{i,t}^{S\,\rho_{k}} \\
+ \gamma_{5,d}^{k} I_{i,t}^{M\,\rho_{k}} + \gamma_{6,d}^{k} I_{i,t}^{T\,\rho_{k}}]^{\frac{1}{\rho_{k}}} e^{\eta_{i,t}^{k}} \qquad k \in \{C,S\}$$
(1)

where  $A_d^k$  is a factor-neutral productivity parameter and  $\rho_k \in (-\infty, 1]$  deter-

<sup>&</sup>lt;sup>7</sup>Note that because the mother is the main caregiver in most families, we focus on her skills as those that are most likely to influence the child's development.

mines the elasticity of substitution, given by  $1/(1-\rho_k)$ , between the inputs affecting the accumulation of skill k. Under such parameterization, as  $\rho_k \to -\infty$ , the inputs become perfect complements. As  $\rho_k \to 1$ , the inputs become perfect substitutes. Notice that we let all the parameters of the production function, except the elasticity of substitution  $\rho_k$ , be a function of the intervention. This choice is dictated by our interpretation of how the intervention could have generated the impacts documented above.

First, the intervention could have changed the parameters of the production function that determine productivity. For example, by providing information about good parenting practices, the intervention could have increased the quality of the investments. In the framework above, this could be reflected by a shift in the factor-neutral productivity parameter  $A_d^k$  between the treated and the non-treated group or a shift in particular share parameters  $\gamma_{j,d}^k$ . Second, the intervention could have affected  $I_{i,t}^M$  and  $I_{i,t}^T$ , the level of material and time investments that parents make, as suggested by the results presented in Table 1. For this reason we will let the parameters of the investment function, which we describe below, be a function of the intervention. In addition, the intervention could also have affected mothers' skills, for instance by improving self-esteem and reducing depression, or maternal knowledge as measured by the Knowledge of Infant Development Inventory (KIDI). Although we checked for these impacts, we did not detect any differences in measures of maternal human capital between control and treated after the intervention, so going forward we assume this mechanism away. From now on, we will therefore assume that mother's human capital is time-invariant and denote her cognitive and socio-emotional skills by  $P_i^C$  and  $P_i^S$ .

A few other features of the production function should be noted. First, all the parameters are specific to a particular skill, so the productivity parameter, the share parameters and the elasticity substitution can differ between the production function of the cognitive skills and that of the socio-emotional skills. Second, the CES functional form provides a great level of flexibility in that it allows the degree of substitutability between the various inputs of the production function to be determined by the data and to range from perfect substitutes to perfect complements. One well-known limitation of the CES functional form is that it imposes the same elasticity of substitution between any two inputs. This could, of course, be alleviated by estimating more general production functions, and in preliminary work we experimented with nested CES production functions. We could not reject the CES functional form however and so we maintain this functional form assumption throughout the application.

There are two main challenges to identifying and estimating the production functions outlined above. The first is related to the fact that parents choose investment in children, so that investments are likely to be correlated with the unobserved shock  $\eta_{i,t}^k$ . The second issue is that children's skills, as well as parental investments and maternal skills, could be measured with error in the data. We discuss how we tackle these two issues in the sub-sections below.

#### **3.2** Accounting for the endogeneity of parental investments

The first issue that complicates the identification of the production function outlined above is the possibility that parental investments are endogenous, which would arise if  $E(\eta_{i,t}^k|I_{i,t}) \neq 0$ . There are two main reasons why parental investments might be correlated with the unobserved shock affecting the accumulation of human capital. First, parental investments might be correlated with omitted inputs in the production function of the child's skills. Second, parental investments might respond to unobserved, time-varying shocks in order to compensate or reinforce their effects on child development. Consider, for example, the case of a child who is suddenly affected by a negative shock, such as an illness, which is unobserved to the econometrician but perceived by the parents as delaying the child's development. As a result of this shock, parents might decide to invest in their child's development more than they would have otherwise. This parental response would create a negative correlation between parental investments and the unobserved shock  $\eta_{i,t}^k$  affecting the development of skills of type k, which would lead the estimate of the effect of parental investments on future skill to be downward biased. Alternative assumptions about preferences and technologies (or technologies as perceived by the parents) can create different patterns of correlations between shocks and investment and, therefore, introduce different types of biases.

Endogeneity, of course, arises because parents choose investment in children to maximize some objective function taking into account the technology of human capital accumulation, the costs of investment and the resources available. In such a context, investment choices in any period will depend on initial conditions, on the shocks affecting the child and observed by the parent, on prices and on total resources. Rather than modeling investment choices jointly with the production function and making specific assumptions on taste (which would imply a specific functional form for the investment function), we estimate a reduced form equation that should be interpreted as an approximation of the investment function. We then use this approximation to implement a control function approach in the estimation of the production function for human capital and, therefore, control for the endogeneity of investment.

For identification, our approach requires that some variables that determine investment choices do not enter the production function directly. A natural candidate would be the intervention we described above, as it was allocated randomly across villages. However, as we want to test whether the intervention changed the parameters of the production function, we cannot use it as an exclusion restriction and need additional variables for valid identification. Moreover, as we model separately different forms of investment, the intervention alone would not be sufficient.

Following Cunha, Heckman, and Schennach (2010), we assume that, for each factor k, the error term  $\eta_{i,t}^k$  can be decomposed into two components,  $\pi_{i,t}^k$ and  $v_{i,t}^k$ . The production function can then be re-written in logs as:

$$\ln(\theta_{i,t+1}^{k}) = \frac{1}{\rho_{k}} \ln[\gamma_{1,d}^{k} \theta_{i,t}^{C\,\rho_{k}} + \gamma_{2,d}^{k} \theta_{i,t}^{S\,\rho_{k}} + \gamma_{3,d}^{k} P_{i}^{C\,\rho_{k}} + \gamma_{4,d}^{k} P_{i}^{S\,\rho_{k}} + \gamma_{5,d}^{k} I_{i,t}^{M\,\rho_{k}} + \gamma_{6,d}^{k} I_{i,t}^{T\,\rho_{k}}] + \ln(A_{d}^{k}) + \delta^{k} \pi_{i,t}^{k} + \upsilon_{i,t}^{k}, \qquad k = \{C, S\}$$

$$(2)$$

Both  $\pi_t^k$  and  $v_{k,t}$  are assumed to be distributed independently across children. However,  $\pi_t^k$  is assumed to be realized before parents make investment choices and therefore can influence their choices, whereas  $v_{i,t}^k$  is realized after parents make investment choices. The goal of the control function approach is to recover a consistent estimate of  $\pi_{i,t}^k$ , so that it can be controlled for when estimating the production function. This in turn requires estimating a model of investment.

As discussed above, we do not derive explicit investment functions from a complete structural model. Instead, we specify an approximation to non-linear investment functions as log-linear equations in initial conditions, maternal skills and the variables  $Z_{i,t}$ , representing resources and prices:

$$\ln(I_{i,t}^{\tau}) = \lambda_{d,0}^{\tau} + \lambda_{d,1}^{\tau} \ln(\theta_{i,t}^{C}) + \lambda_{d,2}^{\tau} \ln(\theta_{i,t}^{N}) + \lambda_{d,3}^{\tau} \ln(P_{i}^{C}) + \lambda_{d,4}^{\tau} \ln(P_{i}^{S}) + \lambda_{d,5}^{\tau} \ln(Z_{i,t}) + u_{i,t}^{\tau}, \qquad \tau = \{M, T\}$$
(3)

where  $u_{i,t}^{\tau}$  is a linear combination of  $\pi_{i,t}^{C}$  and  $\pi_{i,t}^{S}$ . Note that all parameters of the investment functions, including the intercept, are allowed to vary between the treated and non-treated groups. This reflects the possibility, discussed above, that the intervention changed parental investment strategies.

Once the parameters of the investment functions are estimated, we recover  $\hat{u}_{i,t}^T$  and  $\hat{u}_{i,t}^M$ , the estimated residuals from the investment equations (3), which we include as regressors when estimating the production functions:

$$\ln(\theta_{i,t+1}^{k}) = \frac{1}{\rho_{k}} \ln[\gamma_{1,d}^{k} \theta_{i,t}^{C\,\rho_{k}} + \gamma_{2,d}^{k} \theta_{i,t}^{S\,\rho_{k}} + \gamma_{3,d}^{k} P_{i,t}^{C\,\rho_{k}} + \gamma_{4,d}^{k} P_{i,t}^{S\,\rho_{k}} + \gamma_{5,d}^{k} I_{i,t}^{M\,\rho_{k}} + \gamma_{6,d}^{k} I_{i,t}^{T\,\rho_{k}}] + \ln(A_{d}^{k}) + \phi_{M}^{k} \hat{u}_{i,t}^{M} + \phi_{T}^{k} \hat{u}_{i,t}^{T} + v_{i,t}^{k}, \qquad k = C, N$$

$$(4)$$

Identification of the parameters of the production function rests on the as-

sumption that the disturbances  $\pi_{i,t}^k$  and  $v_{i,t}^k$  in equation (2) are independent of  $Z_{i,t}$  and that there are at least as many exclusion restrictions – variables that affect the technology of skill formation only through the investment process – as there are endogenous variables. Economic theory suggests that variables that exogenously shift the household's resources might be valid exclusion restrictions, since they impact parental investment decisions through the budget constraint without entering directly the production function. In this spirit, we use average male and female wages in the child's village, household's wealth at baseline, and an indicator for whether the mother is married as variables that determine resources but do not enter the production function explicitly. These variables are valid exclusion restrictions insofar as, conditional on the child's skills at baseline and maternal human capital, they are orthogonal to  $\pi_t^k$ . We believe that this assumption is likely to hold, as we control for a multitude of child and parents' characteristics through the latent factors.

One of the variables we use as a determinant of investment that does not enter the production function is household wealth. One could argue that household wealth is endogenous to unobserved shocks affecting the child. Indeed, going back to our example above, it is possible that parents of a sick child decide to work more and increase their wealth in order to bolster the care they can provide him or her. Our strategy is less likely to suffer from this caveat, however, because we use baseline measures of wealth that should precede any shocks occurring between t and t+1. Moreover, excluding wealth from the investment functions does not change the point estimates but only affects precision.

### 3.3 Measurement of skills and investments

As described in section 2.2, the data contains multiple measures of the inputs and outputs of the production functions specified above. These measures are likely to proxy a lower-dimensional vector of skills and investment, but to do so with some error. In order to deal with this issue, we follow Cunha, Heckman, and Schennach (2010) in using latent factor models and we estimate the joint distribution of error-ridden latent factors measuring children's skills at baseline  $(\theta_t^C \text{ and } \theta_t^S)$ , children's skills at follow-up  $(\theta_{t+1}^C \text{ and } \theta_{t+1}^S)$ , mother's skills  $(P^C$ and  $P^S)$  and parental investments  $(I_t^M \text{ and } I_t^T)$ , where we keep the individual subscript *i* implicit for notational simplicity. Our specific estimation approach follows the procedure developed in Attanasio, Meghir, and Nix (2015) and is detailed in the next section.

Suppose we have  $\mathcal{M}_{k,t}^1$  measures of child's skills of type k ( $k \in \{C, S\}$ ) at time t. We also have  $\mathcal{M}_k^2$  measures of maternal skills of type k ( $k \in \{C, S\}$ ) and  $\mathcal{M}_{\tau,t}^3$  measures of parental investments of type  $\tau$  ( $\tau = \{M, T\}$ ) at t.<sup>8</sup> Let  $m_{k,t,j}^1$  denote the *j*th measure of child's skill of type k at t,  $m_{k,j}^2$  the *j*th measure of mother's skill of type k, and  $m_{\tau,t,j}^3$  the *j*th measure of parental investment of type  $\tau$  at t. As is common in the psychometric literature, we assume a dedicated measurement system, that is one in which each measure only proxies one factor (Gorusch, 1983, 2003).<sup>9</sup> Assuming each measure is

<sup>&</sup>lt;sup>8</sup>The measures of maternal skills are not indexed by time because we have assumed they are time-invariant.

<sup>&</sup>lt;sup>9</sup>This assumption is not necessary for identification, but we choose to specify a dedicated measurement system so as to make the interpretation of the latent factors more transparent. As described in Appendix C, we find clear support in the data for such a system.

additively separable in the (log) of the latent factor it proxies,<sup>10</sup> we can write:

$$m_{k,t,j}^{1} = \mu_{k,t,j}^{1} + \alpha_{k,t,j}^{1} \ln \theta_{t}^{k} + \epsilon_{k,t,j}^{1}$$
(5)

$$m_{k,j}^2 = \mu_{k,j}^2 + \alpha_{k,j}^2 \ln P^k + \epsilon_{k,j}^2$$
(6)

$$m_{\tau,t,j}^{3} = \mu_{\tau,t,j}^{3} + \alpha_{\tau,t,j}^{3} \ln I_{t}^{\tau} + \epsilon_{\tau,t,j}^{3}$$
(7)

where the terms  $\mu_{k,t,j}^1, \mu_{k,j}^2$  and  $\mu_{\tau,t,j}^3$  are intercepts, the terms  $\alpha_{k,t,j}^1, \alpha_{k,j}^2$  and  $\alpha_{\tau,t,j}^3$  are factor loadings, and the terms  $\epsilon_{k,t,j}^1, \epsilon_{k,j}^2$  and  $\epsilon_{\tau,t,j}^3$  are measurement errors. Note that the latent factors can be freely correlated with each other.

An important specificity of our application of latent factors models is that we consider an intervention and aim to capture its effect on the entire distribution of latent factors. To do so, we allow the joint distribution of the latent factors to be completely different between the two treatment states  $(d = \{0, 1\})$ . In contrast, we assume that the intercepts, factor loadings and measurement errors are invariant across states. These assumptions imply that any difference in the distribution of measures between the control and treated groups result from differences in the distribution of the latent factors and not from differences in the measurement system for those factors. As discussed in Heckman, Pinto, and Savelyev (2013), these assumptions are sufficient but not necessary for identification. We maintain them in our application because they restrict the number of free parameters and lead to improvements in efficiency. Additionally, because the treatment was randomized successfully, there is no reason to think that these parameters should vary across groups.

 $<sup>^{10}</sup>$ We specify the measurement equation such that measures proxy the log of a latent factor so that latent factors only take positive values.

Because the latent factors are unobserved, identification of factor models requires normalizations to set their scale and location (Anderson and Rubin, 1956). We set the scale of the factors by setting the factor loading of the first measure of each latent factor to 1, that is:  $\alpha_{k,t,1}^1 = \alpha_{k,1}^2 = \alpha_{\tau,t,1}^3 = 1$ ,  $\forall t, \tau = \{M, T\}$  and  $k = \{C, S\}$ . We set the location of the factors by fixing the mean of the latent factors to 0 in one group. Without loss of generality, we do so in the control group (d = 0) and allow the latent factor means of the treated group to be freely estimated.<sup>11</sup>

Under these normalizations, Heckman, Pinto, and Savelyev (2013) show that identification of the system is guaranteed as long as we have at least three measures dedicated to each factor under the assumptions that the measurement error is independent across measures and from the latent factors and that  $E(\epsilon_{k,t,j}^1) = E(\epsilon_{k,j}^2) = E(\epsilon_{\tau,t,j}^3) = 0$  for  $j \in \{1, \ldots, \mathcal{M}_{k,t}^1; 1, \ldots, \mathcal{M}_k^2; 1, \ldots, \mathcal{M}_{\tau,t}^3\},$  $\forall t, k \in \{C, S\}$  and  $\tau \in \{M, T\}.$ 

Note that some of these assumptions could be relaxed (Carneiro, Hansen, and Heckman, 2003; Cunha and Heckman, 2008; Cunha, Heckman, and Schennach, 2010). For instance, the same measure could be allowed to load on several factors, as long as there are some dedicated measures. It would also be possible to allow measurement error to be correlated across measures of the same factor, as long as there was one measure whose measurement error was independent from the measurement error in other measures of the same factor.

<sup>&</sup>lt;sup>11</sup>This normalization is innocuous because, without it, we would identify the difference in factor means between the treatment and control groups, which is exactly the object of interest.

# 4 Estimation of the model

The approach we use to estimate our model is described in detail in Attanasio, Meghir, and Nix (2015) and involves two main stages. In the first, we estimate the joint distribution of the latent factors; in the second, we estimate the parameters of the investment and production functions, using draws from the joint distribution of factors.

#### 4.1 First stage: estimating the joint distribution of latent factors

As mentioned above, using the Kotlarski theorem and its extensions by Carneiro, Hansen, and Heckman (2003), Cunha and Heckman (2008) and Cunha, Heckman, and Schennach (2010), the model is non-parametrically identified. For estimation however, we make some distributional assumptions for the distribution of latent factors and measurement error. In particular, we assume that the latent factors are distributed as a mixture of two joint log-normal distributions. The mixture of log normal distribution represents a flexible way to approximate a generic distribution. In principle, one could allow for a mixture of three or more log normal distributions for even greater flexibility, but in our application, we found the two-type mixture satisfactory. Note that it is important to allow for substantial deviation from normality, as such a functional form would imply linearity (or log-linearity in the case of log normality) of the conditional means and, in turn, restrict the elasticity of substitution among factors (for example, normality of the factors would imply that inputs are perfectly substitutable.)

For notational brevity, we denote from now on the vector of latent factors as

 $\theta = (\theta_{t+1}^C, \theta_t^S, \theta_t^C, \theta_t^S, P^C, P^S, I_t^M, I_t^T)$ . As mentioned above, we allow the joint distribution of  $\theta$  to be fully-specific to each intervention group d ( $d \in \{0, 1\}$ ) so their density function in group s can be written as:

$$p_d(\ln \theta) = \tau_d p_d(\ln \theta^A) + (1 - \tau_d) p_d(\ln \theta^B)$$
(8)

where  $\ln \theta^A \sim N(\mu_d^A, \Sigma_d^A)$  and  $\ln \theta^B \sim N(\mu_d^A, \Sigma_d^A)$  and  $\tau_d$  is the mixture weight. In addition, we assume that the measurement errors are distributed as a joint normal distribution with means 0 and diagonal variance-covariance  $\Sigma_{\epsilon}$ . Notice that an implication of the additive separability of the measurement equations (5) - (7), together with the assumption of log-normality of the factors and of normality of the additive measurement error, is that the joint distribution of measurements is given by a mixture of normals.

We first estimate the parameters of the joint distribution of measurements by maximum likelihood, using the EM algorithm. We then map these parameters into the parameters of the joint distribution of factors, the variances of measurement errors, the factor loadings and the intercepts and obtain estimates of these parameters by minimum distance. We report the relationships between the parameters of the distribution of measurements and those of the distribution of factors and of measurement errors in Appendix C. A more detail treatment of the approach is found in Attanasio, Meghir, and Nix (2015).

# 4.2 Second stage: estimating investment and production functions

Once the joint distribution of latent factors is estimated for each group d, we can estimate the investment functions and the production functions using draws from the estimated distributions as data. Given draws on all the latent factors, we first estimate the log-linear investment functions by ordinary least squares and construct the residuals  $\hat{u}_t^{\tau}$  ( $\tau \in \{M, T\}$ ) that serve as control functions. As mentioned above, we let the parameters of the investment function depend on the intervention, reflecting the fact that the intervention might have changed the way parents behave. Next, we estimate the parameters of the CES production functions by non-linear least squares, including the estimated residuals of the investment functions as additional regressors, as specified in equation (4).

To highlight the bias resulting from failing to account for the endogeneity of investments, we present results with and without control functions. We compute standard errors and confidence intervals using the bootstrap.<sup>12</sup>

## 4.3 Specification of the empirical model

In addition to the factors capturing the child's skills at baseline and mother's skills, in the production function, we include the number of children in the household (as measured at follow-up). This is to allow for the possibility that the presence of siblings affects child development, either because of spillover effects or by reducing the level of attention parents devote to each one of their children in multiple children households. We also include the number of children as a determinant of investment, which we suspect might depend negatively on the number of siblings. Since the number of children in the

<sup>&</sup>lt;sup>12</sup>We draw Q = 1000 bootstrap samples of the original data, accounting for the fact that the data is clustered at the village level, and we apply the estimation procedure described above to each one of the pseudo-sample. For each of the parameters, we then compute the standard deviation of its distribution based on its Q = 1000 bootstrapped values, along with various percentiles to compute the corresponding confidence intervals.

household enters the production function directly, it does not enter our list of exclusion restrictions.

To compute our measure of household wealth, we add to the measurement system described above a set of measures from the baseline survey that proxy an additional latent factor measuring household wealth with error. These measures are described in full in Table 2 below and include indicators of whether the household owns its dwelling, along with various other assets (fridge, car, computer, etc.). The other exclusion restrictions (average male and female wages in the village and whether the mother is married), as well as the variable measuring the number of children in the household, are assumed not to have any measurement error.<sup>13</sup>

# 5 Results

We start by reporting the estimates of the measurement system, followed by the estimates of the investment and production functions. Finally, armed with these parameters, we assess how the model fits the data and how it helps us interpret the impact of the early years intervention we have studied.

<sup>&</sup>lt;sup>13</sup>To estimate the joint distribution of all the data we need to estimate the investment and production functions, we therefore specify a measurement system that comprises of all the measures of child's skills, mother's skills, investments and household wealth, along with these four additional variables. Each of these four variables can be thought of being a function of a latent factor and a measurement error term (following the same structure as equation (5) for example), but in their case, the variance of measurement error is 0 and the associated factor loading is 1. With respect to male and female wages in the village, we took an average of male and females wages reported by members of the sample and in doing so rid the average of the measurement error possibly contained in individual observations.

		% Si	gnal			% Si	gnal
Factor	Measures	Controls	Treated	Factor	Measures	Controls	Treated
	Bayley Cognitive	%9 <i>L</i>	% LL		Number of different play materials	%96	97%
Child's	Bayley Receptive Language	71%	72%	Matorial	Number of colouring books	44%	46%
cognitive	Bayley Expressive Language	78%	%6L	INIALEI IAI inviectmente	Number of toys bought	65%	67%
skills	Bayley Fine Motor	55%	57%		Number of toys that require movement	73%	75%
(t+I)	Mac Arthur-Bates Vocabulary	55%	56%		Number of toys to learn shapes	73%	75%
	Mac Arthur-Bates Complex Sentences	38%	39%		Number of different play activities	95%	98%
CLEI JIS	Bayley Cognitive*	74%	%L9	E	Times told a story to child in last 3 days	67%	83%
Cmid S comitive	Bayley Receptive Language*	80%	74%	invectments	Times read to child in last 3 days	70%	85%
cognuve	Bayley Expressive Language*	80%	73%		Times played with child and toys in last 3 days	64%	81%
SKUIIS (+)	Bayley Fine Motor*	68%	60%		Times labelled things to child in last 3 days	65%	82%
(1)	Mac Arthur-Bates Vocabulary*	43%	35%		Mothers' years of education*	64%	63%
	Bates Difficult sub-scale (-)	%69	%L9	Mother's	Mother's vocabulary	70%	%69
Child's socio-	Bates Unsociable sub-scale (-)	21%	20%	cognitive	Number of books for adults in the house*	40%	39%
emotional	Bates Unstoppable sub-scale (-)	62%	%09	skills	Number of magazines and newspapers	18%	17%
skills $(t+I)$	Rothbart Inhibitory Control sub-scale	70%	68%		Raven's score ("IQ") **	60%	59%
	Rothbart Attention sub-scale	25%	24%		Did you feel depressed? (-)	42%	46%
	Bates Difficult factor* (-)	67%	72%		Bothered by what usually don't bother you? (-)	28%	32%
CIIIId S SOCIO- emotional	Bates Unsociable factor* (-)	19%	23%	Mathanla	Had trouble keeping mind on doing? (-)	35%	38%
ebille (+)	Bates Unadaptable* (-)	34%	40%		Felt everything you did was an effort? (-)	31%	34%
(1) SITTAS	Bates Unstoppable* (-)	23%	28%	socio- amotional	Did you feel fearful? (-)	24%	27%
	Owns a fridge	39%	40%	ekille	Did you sleep was restless? (-)	30%	34%
	Owns a car	6%	%9	CITING	Did you feel happy? (-)	13%	15%
	Owns a computer	34%	35%		How often did you feel lonely last week? (-)	31%	35%
Wealth	Owns a blender	28%	29%		Did you feel you couldn't get going? (-)	39%	42%
	Own a washing maching	8%	8%				
	Owns dwelling	11%	11%				
	Owns a radio	12%	13%				
	Ows a TV	32%	33%				
Vote: This tal	ble shows the measures allowed to load o	n each lat	cent factor,	as well as the	fraction of the variance in each measure th	tat is expl	ained

Table 2: Measurement system and signal-to-noise ratio for each measures

by the variance in signal, for the control and treatment groups separately. Measures followed by an asterisks (\*) were collected at baseline and The sign (-) following measures of the child's non-cognitive skill and the mother's non-cognitive skill indicates that the scoring on these measures was reversed so that the corresponding latent factor is such that a higher score means a higher level of non-cognitive skill. measures followed by two asterisks (\*\*) were collected at follow-up II. All other measures were collected at follow-up I at the end of the intervention.

### 5.1 The measurement system and the distribution of factors

Table 2 describes the specification of the measurement system, which underlies the estimates of the production function, that is the set of variables used as measures of each latent factor. To arrive at this specification, we performed an exploratory factor analysis of the data that helped us to determine the number of factors that could be extracted from the data and to allocate measures to particular factors. The steps and results of this exploratory factor analysis are discussed in detail in Appendix C.

As mentioned above, identification requires that at least one measure for each factor is conditionally independent of the other measures for the same factor. In our case, this assumption can be justified by the fact that some developmental outcome variables are based on child level observations and are collected by a trained psychologist in community centers, while others are based on maternal reports and are collected in the home (on a different day) by an interviewer. The independence of measurement errors is probably not a far fetched assumption in such a context.

From the first stage of the estimation procedure, we obtained estimates of the measurement system, i.e. estimates of the mean and variance-covariance matrix of the latent factors for each group  $d = \{0, 1\}$ , estimates of the factor loadings and of the variances of the measurement error.<sup>14</sup> All these parameter estimates are reported in Appendix C.<sup>15</sup> Using these estimates, it is possible to

<sup>&</sup>lt;sup>14</sup>We standardized all measures with respect to their mean in the control group, so we bypass the estimation of the intercepts.

<sup>&</sup>lt;sup>15</sup>More precisely, Appendix Table A3 reports the estimates of the factor loadings in each measurement equation. Appendix Table A4 reports the estimates of the means of the latent factor distributions for the treated households relative to the means of the latent factor

assess the extent of measurement error in each measure of skill and investment in the data. Simplifying the notation, recall that the *j*th measure of latent factor  $\theta$  can be written as:

$$m_j = \mu_j + \alpha_j \theta + \epsilon_j$$

Given this functional form and our maintained assumption that the factors are independent from the measurement error, the variance of  $m_j$  can be decomposed as:  $Var(m_j) = \alpha_j^2 Var(\theta) + Var(\epsilon_j)$ . We can then calculate the fractions of  $Var(m_j)$  due to signal  $(s^{\theta})$  and noise  $(s^{\epsilon})$  as follows:

$$s_j^{\theta} = \frac{\alpha_j^2 Var(\theta)}{\alpha_j^2 Var(\theta) + Var(\epsilon_j)} \qquad \text{and} \qquad s_j^{\epsilon} = 1 - s_j^{\theta} \tag{9}$$

We report estimates of  $s^{\theta}$  and  $s^{\epsilon}$  in Table 2 for the control and treated groups. Clearly, there is much variation in the amount of information contained in each measure of the same factor. For example, 76% of the variance in Bayley Cognitive is due to signal, whereas only 38% of the variance in Mac Arthur-Bates Complex Sentences is due to signal. However, no measure is close to having 100% of its variance accounted for by signal, which emphasizes the importance of accounting for measurement error through the latent factor model.<sup>16</sup>

distributions for the control households (which is normalized to 0). Appendix Table A5 reports the estimates of the variance of and correlation between the latent factors for the treated and control households separately.

<sup>&</sup>lt;sup>16</sup>There is little difference in the signal-to-noise ratio of measures between the control and treated groups. Under our assumption of invariant factor loadings and measurement error between the control and treated groups, differences in signal-to-noise ratios could only come from differences in the variances of the latent factors between the two groups. Appendix Table A5 shows that these are minimal.



Figure 1: Kernel densities of latent factors

*Note*: These kernel densities are constructed using 10,000 draws from the estimated joint distribution of latent factors for the control group and for the treated group.

Having identified the entire distribution of factors for each group, we can study whether the intervention has changed the entire shape of these distributions, in addition to their means. In Figure 1, we plot the estimated Kernel densities of some of the factors. The first two panels show the distribution, in treatment and control villages, of cognitive and socio-emotional skills at baseline. These first two pictures confirm that our sample is substantially balanced. The following two panels depict the distribution of cognitive and socio-emotional factors at follow-up. In the case of cognitive factors we see that the shift in the mean reported in Appendix Table A4 reflects a shift in the entire distribution. For socio-emotional factors, however, the shift occurs mainly for children below the median.

Finally, in the last two panels, we notice a strong shift to the right of both the material and time investment factors. This suggests that at least part of the impact of the intervention is likely to have been driven by increases in both time and materials devoted by parents to the upbringing of their children.

# 5.2 Estimates of the investment functions

In Table 3, we present estimates of the investment equations. The first column presents the equation for material investments and the second column for time investments. As far as we know, our paper is unique in distinguishing between material and time investments in the context of estimating non-linear technologies of skill formation. Note that the results reported in Table 3 exclude interactions of the treatment parameter with the remaining variables. In earlier versions we found such interactions to be insignificant, i.e. the shift
	Log of material Log of time		
	investments	investments	
Constant	0.001	0.004	
	(0.016)	(0.015)	
	[-0.025,0.027]	[-0.02,0.028]	
Treatment dummy	0.248	0.361	
	(0.073)	(0.065)	
	[0.115,0.349]	[0.235,0.451]	
Log of child's cognitive skills at t	0.141	0.116	
	(0.061)	(0.057)	
	[0.032,0.231]	[0.007,0.198]	
Log of child's socio-emotional skills at t	-0.008	0.031	
	(0.058)	(0.056)	
	[-0.105,0.084]	[-0.053,0.13]	
Log of mother's cognitive skills	0.668	0.462	
	(0.082)	(0.079)	
	[0.54,0.815]	[0.317,0.573]	
Log of mother's socio-emotional skills	-0.120	-0.310	
	(0.074)	(0.103)	
	[-0.25,-0.001]	[-0.474,-0.123]	
Log of wealth at <i>t</i>	0.081	-0.086	
	(0.071)	(0.090)	
	[-0.019,0.217]	[-0.231,0.06]	
Mother is married at $t+1$	0.126	0.115	
	(0.027)	(0.027)	
	[0.075,0.164]	[0.066,0.155]	
Log of number of children at $t+1$	-0.096	-0.090	
	(0.033)	(0.033)	
	[-0.146,-0.04]	[-0.151,-0.041]	
Log of average male wages in village at $t+1$	0.075	-0.026	
	(0.041)	(0.044)	
	[-0.007,0.117]	[-0.106,0.039]	
Log of average female wages in village at $t+1$	0.004	0.033	
	(0.038)	(0.031)	
	[-0.077,0.054]	[-0.013,0.088]	

Table 3: Estimates of the log-linear investment function

*Note*: Standard errors in parentheses and 90% confidence intervals in brackets are obtained using the non-parametric bootstrap described in Section 4. Appendix B provides a detailed description of the variables used to measure each latent factor.

in investment seems to have been uniform across groups with differing backgrounds.<sup>17</sup>

The first striking result is the impact of treatment on investments: it increases resources by 25% and time by 36% and both effects are highly significant. Thus, the intervention increased the time and the resources that parents provide to children. Referring back to the measurement system (Table 2), it is worthwhile noting that the time inputs are measured in a way that are targeted to child educational activities, such as the number of times an adult read to the child in the last three days. In other words, they do not refer simply to time spent with the child, but to interactions that promote development. Similarly, material investments refer to particular types of toys and play materials. Importantly, our estimates of the impact of the intervention on investments are uniquely driven by the experimental design and do not require any of the assumptions necessary for the identification of the production functions.

Turning now to the other regressors, we find that both time and material investments increase with the child's cognitive skills, but socio-emotional skills have no impact on investments, at least at the very young ages we are considering. The elasticity of both material and time investments with respect to maternal cognition is very high and particularly so for the former; however mother's socio-emotional skills have no significant effect. Married mothers in-

<sup>&</sup>lt;sup>17</sup>Appendix Table A6 reports the estimates of the investment equation in which we interact all the regressors with the treatment dummy. We test the joint significance of the interaction terms and find that we cannot reject that all the interactions are equal to 0 for both material and time investments at conventional levels of significance (the *p*-value for the material investment equation is 0.577 and the *p*-value for the time investment equation is 0.667). Appendix Table A7 reports the estimates of the production functions when the control functions are derived from the fully interacted investment function. We observe no major difference between these results and the ones based on the non-interacted investment function reported in Tables 4 and 5.

vest more time and more materials, but the overall number of children reduces investments. We find no effect of wealth on investments, probably because our sample is quite homogeneously poor. However, we do find that more is invested in children in terms of materials when male wages are higher in the village, which we interpret as an income effect. Average female wages in the village, which could reflect the opportunity cost of time for the mother as well as her access to resources, have no effect.

Of the above variables, wealth, marital status, and male and female wages at the village level are excluded from the production function and serve as instruments. We test for the joint significance of the exclusion restrictions in the material and time investment equations. The *p*-value from a F-test performed separately on each equation is 0.008 for the material investment equation and 0.041 for the time investment equation. The *p*-value of a test that the rank of the coefficient matrix of the excluded variables across both investment equations is less than 2 has a *p*-value of 0.035.<sup>18</sup> The fact that some of the coefficients on the individual variables are not statistically different from zero is probably driven by collinearity and by the limited variation we have (as some of these variables only vary across village). From the high joint significance of our instruments however, we conclude that they are strong enough to identify the impact of investment in the production function for skills, to which we now turn.

<sup>&</sup>lt;sup>18</sup>To perform this rank test, we proceeded as follows. Let *B* be the 2 × 4 matrix of coefficients on the four excluded variables in the two investment equations. We used the non-parametric bootstrap to derive critical values for a test that the smallest eigenvalue of B'B is zero. See Robin and Smith (2000) for a discussion of such tests and Blundell, Duncan, and Meghir (1998) for a similar application.

# 5.3 Estimates of the production functions

Tables 4 and 5 report the estimates of the CES production function. For each skill, we report estimates of the parameters of the production function obtained both ignoring the endogeneity of investments (in the left column) and taking it into account by including the residuals from the investment functions as additional inputs (in the right column), as specified in equation (4). It should be stressed that very limited systematic evidence exists on the evolution and development of children skills at such young ages.

We have experimented with allowing all coefficients to vary with treatment status, but we found no evidence that the production function differs systematically as a result of the intervention. In fact the tests that the coefficients are the same across intervention and control groups, other than total factor productivity A, have p-values of 0.879 and 0.889 for the production functions of cognitive and socio-emotional skills, respectively.<sup>19</sup> Indeed, as Tables 4 and 5 show, even total factor productivity remains unaffected by treatment, except when we treat investments as exogenous in the production function for cognitive skills (first column of Table 4).<sup>20</sup>

The results provide evidence of several important features of skill development. First, we find strong evidence of self-productivity of skills. That is, the current stock of cognitive (socio-emotional) skills strongly affects the

<sup>&</sup>lt;sup>19</sup>Appendix Tables A8 and A9 report the estimates of the production function for each skill type in which we allow all parameters (except the elasticity of substitution) to vary between treated and controls. None of the interactions, except that with the number of children in the household, is significant.

<sup>&</sup>lt;sup>20</sup>We also experimented with a specification of the model where we allowed the child's age at follow-up to enter the investment equations and to interact with the total-factor productivity parameter (A) in the production functions. We did not find evidence that this parameter depended on the child's age however (see Appendix Tables A10 and A11.)

	Without control	With control
	function	function
Child's cognitive skills at t	0.591	0.566
	(0.043)	(0.057)
	[0.527,0.67]	[0.489,0.674]
Child's socio-emotional skills at t	0.03	0.038
	(0.043)	(0.050)
	[-0.037,0.106]	[-0.035,0.126]
Mother's cognitive skills	0.194	0.037
	(0.049)	(0.131)
	[0.107,0.264]	[-0.194,0.223]
Mother's socio-emotional skills	0.06	0.051
	(0.045)	(0.049)
	[-0.016,0.126]	[-0.028,0.127]
Material investments at $t+1$	0.082	0.397
	(0.033)	(0.208)
	[0.036,0.144]	[0.128,0.765]
Time investments at $t+1$	0.008	-0.138
	(0.035)	(0.142)
	[-0.056,0.057]	[-0.421,0.039]
Number of children in household at $t+1$	0.035	0.049
	(0.026)	(0.030)
	[-0.009,0.076]	[0.002,0.1]
Control function for material investments	-	-0.33
		(0.218)
		[-0.715,-0.023]
Control function for time investments	-	0.156
		(0.151)
		[-0.037,0.453]
Complementarity parameter	0.123	0.07
	(0.082)	(0.060)
	[-0.025,0.243]	[-0.032,0.161]
Elasticity of substitution	1.141	1.075
	(0.106)	(0.070)
	[0.976,1.321]	[0.969,1.192]
Productivity parameter (A)	0.984	0.993
	(0.012)	(0.011)
	[0.966,1.005]	[0.972,1.008]
Productivity parameter interacted with treatment	0.1	0.08
	(0.052)	(0.072)
	[0.028,0.198]	[-0.012,0.228]

Table 4: Estimates of the CES production function for cognitive skills

*Note*: Standard errors in parentheses and 90% confidence intervals in brackets are obtained using the non-parametric bootstrap described in Section 4. Appendix B provides a detailed description of the variables used to measure each latent factor.

	Without control	With control
	function	function
Child's cognitive skills at t	0.11	0.122
	(0.044)	(0.059)
	[0.039,0.185]	[0.024,0.222]
Child's socio-emotional skills at t	0.435	0.413
	(0.055)	(0.059)
	[0.374,0.552]	[0.354,0.537]
Mother's cognitive skills	-0.054	0.116
	(0.066)	(0.142)
	[-0.168,0.046]	[-0.201,0.276]
Mother's socio-emotional skills	0.151	0.161
	(0.058)	(0.058)
	[0.047,0.233]	[0.046,0.235]
Material investments at $t+1$	0.14	-0.32
	(0.043)	(0.198)
	[0.079,0.219]	[-0.529,0.108]
Time investments at $t+1$	0.119	0.434
	(0.041)	(0.133)
	[0.043,0.181]	[0.17,0.591]
Number of children in household at $t+1$	0.099	0.073
	(0.026)	(0.027)
	[0.048,0.136]	[0.025,0.113]
Control function for material investments	-	0.477
		(0.204)
		[0.043,0.711]
Control function for time investments	-	-0.336
		(0.136)
		[-0.506,-0.068]
Complementarity parameter	0.049	0.006
	(0.077)	(0.056)
	[-0.085,0.158]	[-0.059,0.12]
Elasticity of substitution	1.051	1.006
	(0.088)	(0.063)
	[0.921,1.187]	[0.944,1.137]
Productivity parameter (A)	0.987	0.992
	(0.016)	(0.012)
	[0.966,1.019]	[0.976,1.014]
Productivity parameter interacted with treatment	-0.015	-0.016
	(0.042)	(0.058)
	[-0.081,0.054]	[-0.115,0.0/3]

Table 5: Estimates of the CES production function for socio-emotional skills

*Note*: Standard errors in parentheses and 90% confidence intervals in brackets are obtained using the non-parametric bootstrap described in Section 4. Appendix B provides a detailed descriptions of the variables used to measure each of the latent factor.

development of future (socio-emotional) cognitive skills. Second, we find evidence of cross-productivity in the production function for socio-emotional skills (Table 5): the current stock of cognitive skills fosters the development of future socio-emotional skills. Although the current stock of socio-emotional skills has a positive coefficient in the production function of cognitive skills, the estimate is not significantly different from zero and we cannot conclude there is cross-productivity in the production of cognitive skills. This result contrasts with that reported by Cunha, Heckman, and Schennach (2010), who find socio-emotional skills to be important for the accumulation of future cognitive skills. Our evidence, however, relates to a different, much earlier, period of the life-cycle.

Our results also provide evidence that parental investments play a crucial role in the development of both cognitive and socio-emotional skills and that they are endogenous. With respect to cognitive skills, the results reveal that only material investments seem to matter; their effect becomes very large and strongly significant, once we control for endogenous investments. With respect to socio-emotional skills, when we treat investments as endogenous, only time investments significantly matter for the accumulation of socio-emotional skills. The point estimate of the share parameter for material investments is negative, but very imprecise. The effect of time investments on future socio-emotional skills is very large and strongly significant.

Investments are also clearly endogenous. The control function in the cognitive production function is highly significant for material investments and including it increases the share parameter for material investments more than fivefold. Similarly the control function for time investments in the corresponding production function is also highly significant, and including it in the production function triples the coefficient on time investments. In both of these cases, the estimate of the coefficient of the relevant control function is negative, suggesting that parents compensate with materials and time when adverse events affect the development of their child. This evidence is important as it speaks to the issue of whether parental investment reinforces or compensates shocks received by children, an issue discussed by a large literature, some of which is surveyed in Almond and Mazumder (2013). Our findings indicate very clearly that investment seems to be compensating rather than reinforcing.

The estimates of the production functions show that mothers' cognitive skills do not seem to have a significant *direct* effect on either children's cognitive or socio-emotional development, once we control for the endogeneity of investments. However, it should be remembered that according to our estimates of the investment function (Table 3), mother's cognitive skills have an important effect on both types of investment and, therefore, an indirect effect on child development. Mother's socio-emotional skills have a significant impact on the development of the child's socio-emotional skills, even after controlling for the endogeneity of investments, but their effect on the child's future cognitive skills is not significantly different from zero.

It is also interesting that the number of siblings improves outcomes for children *conditional* on investments, presumably because of mutual stimulation and interaction. On the other hand, the estimates of the investment function imply that siblings also dilute investments to the subject child, thus creating two counteracting effects.

The last key component of the production function is the complementarity parameter,  $\rho_k$ . In both production functions, it is estimated to be zero, which implies an elasticity of substitution equal to one. That is, the production function for cognitive and socio-emotional skills has a Cobb-Douglas form.

# 5.4 Implications for human capital accumulation and the role of the intervention

The results presented above have interesting implications for human capital accumulation. First, the production function is Cobb-Douglas and hence inputs are complementary. Moreover, baseline skills, investments and, to some extent, maternal skills are important for the development of future skills. Together, these two facts imply that investments are more productive for children with higher levels of early cognition and socio-emotional skills.

We illustrate this point in Figure 2, which plots the marginal product of investments as a function of the child's baseline skills. Consider first Figure 2a. The productivity of material investment in the production function for cognitive skills triples as we move from the lower deciles to the upper deciles of the distribution of baseline cognition. Similarly, time investments are more productive for the accumulation of socio-emotional skills, the higher a child is on the distribution of socio-emotional skills at baseline (Figure 2b). This complementarity illustrates both the importance of targeting low-achieving children early as well as the difficulty of delivering interventions with high returns for them, particularly as time passes.



Figure 2: Complementarity between investments and baseline skills

(a) Marginal product of material investments in the production of cognitive skills

Decile Decile

(b) Marginal product of time investments in the production of socio-emotional skills



Notes: The marginal product of material and time investments in the production function for skill of type k are given by:  $A_d^k \gamma_{5,d}^k I_{i,t}^{M(\rho_k-1)} Q^{(\frac{1-\rho_k}{\rho_k})}$  and  $A_d^k \gamma_{6,d}^k I_{i,t}^{T(\rho_k-1)} Q^{(\frac{1-\rho_k}{\rho_k})}$ , respectively, where  $Q = [\gamma_{1,d}^k (\theta_{i,t}^C)^{\rho_k} + \gamma_{2,d}^k (\theta_{i,t}^S)^{\rho_k} + \gamma_{3,d}^k (P_i^C)^{\rho_k} + \gamma_{4,d}^k (P_i^S)^{\rho_k} + \gamma_{5,d}^k (I_{i,t}^M)^{\rho_k} + \sum_{k=1}^{N} (P_k^{C-1}) Q^{(k-1)} Q^$  $\gamma_{6,d}^K(I_{i,t}^T)^{\rho_k}$ ]. The figures above are constructed by evaluating these objects at different decides of  $\theta_{i,t}^C$  (figures i) and  $\theta_{i,t}^S$  (figures ii) and holding all remaining inputs of the production function at their mean values pooling observations from the control and treatment groups together.

As we mentioned above, the factor-neutral productivity parameter A is not shifted by the treatment indicator, whose coefficient is small and insignificant in either production function. Moreover, when we test whether the intervention has shifted any of the other parameters of the production function, we cannot reject the null of no change. On the other hand, there is a positive and significant shift in parental investments among treated families relative to nontreated families (Figure 1), and on average there is a 25% increase in material investments and 36% increase in time investments among treated households relative to control households. These results imply that the intervention acted exclusively by changing parental investments in materials and time and not by increasing the productivity of investments nor that of any other inputs. Given the evidence we just presented on the role that these investments play in the production function for human capital, it is natural to ask whether this channel can explain all the impact of the intervention.

More formally, we ask what fraction of the difference in skills between the treated and non-treated children can be explained by the change in parental investments induced by the intervention. We can answer this question by using the structure and estimates of the model. In particular, we predict the distributions of future cognitive and socio-emotional skills given the distribution of investments in the treatment group and those of the control group respectively. We then compute the gap between the two for each type of skill. This difference measures how much of the impact of the intervention we can explain using just the change in investments.

Table 6 reports the results from this exercise based on the estimates with

	Gap in log skills betw	Gap in log skills between treated and controls		
	Cognitive skills	Socio-emotional skills		
	at <i>t</i> +1	at <i>t</i> +1		
Data	0.115	0.075		
	(0.056)	(0.045)		
	[0.011,0.195]	[0.007,0.151]		
Model	0.107	0.088		
	(0.058)	(0.047)		
	[0.009,0.195]	[0.005,0.156]		

Table 6: Decomposition of the treatment effects in skills

*Note*: Standard errors in parentheses and 90% confidence intervals in brackets are obtained using the non-parametric bootstrap described in Section 4. Appendix B provides a detailed description of the variables used to measure each latent factor.

endogenous investments. The results indicate that the model predicts the actual gap in log skills between the treated and non-treated children well for both type of skills. Specifically the model predicts 93% of the cognitive improvement and slightly overshoots the socio-emotional improvement among treated children. The differences between the data and the model predictions are not significant. Thus the conclusion is quite clear: the intervention changed parental behavior by inducing more of the appropriate investments but did not change the production function.

An interpretation of this finding is that, as a result of the intervention, parents' behavior did not become more productive in itself, but parents allocated more of their resources on their children, given their beliefs on the effectiveness of these investments. This suggests that a reason for under-investment in child development among poor populations might be a lack of knowledge about the effectiveness of investments, thus implying that one way interventions can be successful is by changing parenting beliefs on the subject.

# 6 Conclusion

In this paper, we estimate the production functions for cognitive and socioemotional skills in early childhood and analyze the mechanisms through which a randomized early childhood intervention in Colombia affected child development. The intervention itself - a psychosocial stimulation program delivered through weekly home visits - significantly improved the cognitive and socioemotional development of children aged between 12 and 24 months at the start of the intervention. Underlying our approach is a model of parental investments in children, constrained by a production function for cognitive and socio-emotional skills.

We make three interrelated contributions. First, we estimate equations for material and time investments depending on both family and child baseline characteristics. Second, we estimate production functions for cognitive and socio-emotional skills that depend on the investments, which are taken to be endogenous. Third, we use this framework to trace out the way the intervention affected the production of children's skills.

In line with the existing literature, we find strong evidence that a child's current stock of skills fosters the development of future skills (self-productivity). We also find evidence that early cognition helps improve socio-emotional skills (cross-productivity). Second, and also in line with the existing literature, we find that current skills, parental investments and maternal human capital are

complementary in the production of future skills. This means, for example, that parental investments are more productive when the child's current skills are higher. Lastly, we find that parental investments greatly matter for the accumulation of cognitive and socio-emotional skills. In particular, material investments seem to matter more for cognitive skills, while time investments seem to matter more for socio-emotional skills. Our paper is the first to estimate a non-linear production function for skills with different types of investments, and our results show that it is a non-trivial extension of the standard model.

When we compare the production function estimates between treatment and control groups, we cannot reject that they are the same. We find that we can attribute the entire effect of the intervention to an increase in time and material resources devoted by parents to their children. In other words, the intervention did not shift the way that skills are produced. Nevertheless, we show that even a shift in investments without a change in the production function succeeded in altering children's outcomes in significant ways. Whether the intervention changed parental behavior fundamentally enough to yield long-lasting impacts is an open question, but one of crucial importance as further follow-up data is collected. The framework developed here surely offers a valuable tool to understand what drives the sustainability of this and other early years interventions and organize the evidence in a systematic way.

# References

- Almond, D. and Currie, J. (2011). Human capital development before age five. In O. Ashenfelter and D. Card, eds., *Handbook of Labor Economics, Volume* 4b. Elsevier Science, Amsterdam.
- Almond, D. and Mazumder, B. (2013). Fetal origins and parental responses. The Annual Review of Economics 5(3):1–20.
- Anderson, T. and Rubin, H. (1956). Statistical inference in factor analysis. In Proceedings of the Third Berkeley Symposium on Mathematical Statistics and Probability, volume 5, 111–150. University of California Press, Berkeley.
- Attanasio, O., Fernandez, C., Fitzsimons, E., Grantham-McGregor, S., Meghir, C., and Rubio-Codina, M. (2014). Using the infrastructure of a Conditional Cash Transfer programme to deliver a scalable integrated early child development programme in Colombia: A cluster randomised controlled trial. Forthcoming in *British Medical Journal* 349:g5785.
- Attanasio, O., Meghir, C., and Nix, E. (2015). Investments in children and the development of cognition and health in India. Mimeo, Yale University.
- Attanasio, O., Meghir, C., and Santiago, A. (2012). Education choices in Mexico: Using a structural model and a randomized experiment to evaluate Progresa. *Review of Economic Studies* 79(1):37–66.
- Bates, J., Freeland, C., and Lounsbury, M. (1979). Measurement of infant difficultness. *Child Development* 50:794–803.
- Bayley, N. (2006). Bayley Scales of Infant and Toddler Development. Harcourt Assessment, San Antonio, TX, 3rd edition.
- Blundell, R., Duncan, A., and Meghir, C. (1998). Estimating labor supply responses using tax reforms. *Econometrica* 66(4):827–761.
- Carneiro, P., Hansen, K., and Heckman, J. (2003). Estimating distribution of counterfactuals with an application to the returns to schooling and measurement of the effect of uncertainty on schooling choice. *International Economic Review* 44(2):361–422.
- Cunha, F. and Heckman, J. (2008). Formulating, identifying and estimating the technology of cognitive and noncognitive skill formation. *Journal of Human Resources* 43(4):738–782.

- Cunha, F., Heckman, J., Lochner, L., and Masterov, D. (2006). Interpreting the evidence on life cycle skill formation. In E. Hanushek, S. Machin, and L. Woessmann, eds., *Handbook of the Economics of Education, Volume 1*. Elsevier Science, Amsterdam.
- Cunha, F., Heckman, J., and Schennach, S. (2010). Estimating the technology of cognitive and non-cognitive skill formation. *Econometrica* 78(3):883–931.
- Duflo, E., Hanna, R., and Ryan, S. P. (2012). Incentives work: Getting teachers to come to school. *American Economic Review* 102(4):1241–78.
- Frongillo, E., Sywulka, S., and Kariger, P. (2003). UNICEF psychosocial care indicators project. Final report to UNICEF. Mimeo, Cornell University.
- Gertler, P., Heckman, J., Pinto, R., Zanolini, A., Vermeerch, C., Walker, S., Chang, S., and Grantham-McGregor, S. (2014). Labor market returns to an early childhood stimulation intervention in Jamaica. Forthcoming, *Science*.
- Gorusch, R. (1983). Factor Analysis. Lawrence Erlbaum Associates, Hillsdale, NJ.
- Gorusch, R. (2003). Factor analysis. In Handbook of psychology: Research methods in psychology, volume 2, 143–164. John Wiely & Sons, Inc., Hoboken, NJ.
- Grantham-McGregor, S., Powell, C., Walker, S., and Himes, J. (1991). Nutritional supplementation, psychosocial stimulation, and mental development of stunted children: the Jamaican study. *Lancet* 338(758):1–5.
- Heckman, J., Moon, S., Pinto, R., Savelyev, P., and Yavitz, A. (2010). Analyzing social experiments as implemented: A reexamination of the evidence from the HighScope Perry Preschool Program. *Journal of Quantitative Economics* 1:1–46.
- Heckman, J., Pinto, R., and Savelyev, P. (2013). Understanding the mechanisms through which an influential early childhood program boosted adult outcomes. *American Economic Review* 103:2052–2086.
- Helmers, C. and Patnam, M. (2011). The formation and evolution of childhood skill acquisition: Evidence from India. *Journal of Development Economics* 95(2):252–266.

- Jackson-Maldonado, D., Marchman, V., and Fernald, L. (2012). Short form versions of the Spanish MacArthur-Bates Communicative Development Inventories. In Applied Psycholinguistics. Cambridge University Press, Cambridge.
- Knudsen, E. (2004). Sensitive periods in the development of the brain and behavior. Journal of Cognitive Neuroscience 16(1):1412–1425.
- Knudsen, E., Heckman, J., Cameron, J., and Shonkoff, J. (2006). Economic, neurobiological and behavioral perspectives on building America's future workforce. *Proceedings of the National Academy of Science* 103(27):10155– 10162.
- McGregor, S. G., Cheung, Y., Cueto, S., Glewwe, P., Richter, L., and Strupp, B. (2007). Developmental potential in the first 5 years for children in developing countries. *The Lancet* 369(9555):60–70.
- Putnam, S. P., Gartstein, M., and Rothbart, M. K. (2006). Measurement of fine-grained aspects of toddler temperament: The Early Childhood Behavior Questionnaire. *Infant Behavior and Development* 29(3):386–401.
- Robin, J. and Smith, R. (2000). Tests of rank. *Econometric Theory* 16:151–175.
- Todd, P. and Wolpin, K. (2006). Using experimental data to validate a dynamic behavioral model of child schooling and fertility: Assessing the impact of a school subsidy program in Mexico. American Economic Review 96(5):1384–1417.
- Walker, S., Chang, S., C.Powell, E.Simonoff, and Grantham-McGregor, S. (2006). Effects of psychosocial stimulation and dietary supplementation in early childhood on psychosocial functioning in late adolescence: follow-up of randomized controlled trial. *British Medical Journal* 333(7566):472.
- Walker, S., Chang, S., Vera-Hernandez, M., and Grantham-McGregor, S. (2011). Early childhood stimulation benefits adult competence and reduces violent behavior. *Pediatrics* 127(5):849–857.

# Estimating the Production Function for Human Capital: Results from a Randomized Control Trial in Colombia

Orazio Attanasio, Sarah Cattan, Emla Fitzsimons, Costas Meghir, and Marta Rubio-Codina<sup>\*</sup>

# APPENDIX FOR ONLINE PUBLICATION

February 7, 2015

<sup>\*</sup>Attanasio: University College London and Institute for Fiscal Studies (o.attanasio@ucl.ac.uk). Cattan: Institute for Fiscal Studies (sarah\_c@ifs.org.uk). Fitzsimons: UCL Institute of Education and Institute for Fiscal Studies (e.fitzsimons@ioe.ac.uk). Meghir: Yale University, NBER and Institute for Fiscal Studies (c.meghir@yale.edu). Rubio-Codina: Institute for Fiscal Studies and Inter-American Development Bank (marta\_r@ifs.org.uk). We thank participants at the NBER Summer Institute, Barcelona GSE Summer Forum and Montreal CIREQ Applied Microeconomics on Fertility and Child Development and seminars at Stanford University, University of Chicago, Oxford University, Cornell University, Bristol University and the Institute for Fiscal Studies for their comments. We are grateful to the Economic and Social Research Council (Grant ES/G015953/1), the Inter-American Development Bank, the International Growth Centre, and the World Bank for funding the intervention and data collection. Some of this research was financed by the European Research Council's Advanced Grant 249612 and by the Grand Challenges Canada Prime Award 0072-03 (sub-award reference number 560450). Sarah Cattan gratefully acknowledges financial assistance from the British Academy Postdoctoral Fellowship pf140104, as well as from the European Research Council's Grant Agreement No. 240910. Costas Meghir thanks the Cowles foundation and the ISPS at Yale for financial assistance. All errors are the responsibility of the authors.

# A The Intervention

This appendix provides a detailed description of the intervention, its implementation and roll-out. The material in this section draws heavily on Attanasio et al. (2014).

# A.1 Description of the Intervention

The integrated early child development intervention was designed so that it could be delivered by local people, readily identifiable through administrative infrastructures of social welfare systems (the conditional cash transfer program Familias en Accion (FeA) in our case). The intervention included psychosocial stimulation on its own, micronutrient supplementation on its own, and both combined. Each of these arms had 24 clusters (municipalities).

# A.1.1 Psychosocial Stimulation

The psychosocial stimulation component was inspired by and based on the Jamaican home visiting model (Walker et al 2011), the overarching aim of which is to facilitate developmentally appropriate learning activities between mothers (primary caregivers) and their children through demonstration of play activities centred around daily routines. Play activities draw on resources in the home, low cost home-made toys and the intervention toy kit. The toy kit included picture books, naming plates, conversation scenes, puzzles, lotteries, and blocks. Play activities followed steps aimed towards children's gradual mastery of a learning objective: (1) child observes play activity (modeling),

(2) mother and child do the activity together, (3) child attempts the play activity on his/her own, (4) mother prompts naming and verbalisation of objects and actions linked to the play activity, (5) the developmental level of play activity is adjusted by increasing or decreasing the challenge based on the child's performance (scaffolding). Throughout the play activities, mothers are encouraged to provide children with contingent positive reinforcement for progress toward the learning goals (praise) and to follow the child's interest.

Materials for the stimulation component were adapted from the Jamaican intervention guide to the local cultural context and to the average educational level of home visitors and program beneficiaries. Such adaptations included (1) inclusion of local songs and rhymes; (2) modification of the home-made toys instruction manual to use local recyclable materials, (3) incorporation into the the intervention toy kit of culturally relevant pictures, scenes and objects, (4) re-organization of the psychosocial stimulation guide in weekly instruction cards for specific age groups, and (5) ordering of play materials, via an index, by developmental stage to facilitate the scaffolding of activities.

Home visits lasted approximately one hour. Home visits took place with the child's biological mother or primary caregiver. Other adults in the household, where present, were also encouraged to participate in the home visit. In advance of the visit, the home visitor selected the weekly instruction card from the psychosocial stimulation guide according to the appropriate developmental level of the target child, and prepared the toys and materials for the visit. The home visit had three parts. At the beginning, the home visitor did an informal assessment of the child's progress in the sequence of play activities, by asking about the play activities the mother and child practiced during the previous week and identifying where challenges arose (or on the contrary, where activities were overly simplistic). In the second and main part of the visit, new play activities were shown and practised. At the end, the mother was prompted to summarize the play activities and agree a plan on how to practice them over the following week.

# A.1.2 Micronutrient Supplementation

The micronutrient supplementation component consisted of micronutrient supplementation in the form of sprinkles (encapsulated micronutrients). Each single-dose sachet contains 12.5 mg iron, 5 mg zinc, vitamin A 300 mg RE, 160mg g folic acid and 30 mg vitamin C and each displayed a pictorial representation of use. In addition to the fortnightly provision of sprinkles, participating families received a booklet with detailed instructions for use and storage and daily record forms to track use. Families were provided with enough sachets for all children below six years of age to prevent sharing with siblings.

# A.2 Implementation

We obtained institutional cooperation from the National Director of FeA, the coordinator of the Strategic Planning and Monitoring Unit of FeA, and FeA administrative staff at the municipality level. From the latter, we obtained rosters of female community representatives (Madres Lideres) and selected three female community representatives in each of the 96 target municipalities.

In municipalities assigned to receive stimulation, 63% of selected female

community leaders took on the role of home visitor. The remainder declined due to other work commitments or not meeting the minimum reading comprehension criteria (established using a short reading comprehension test designed by the study data collection team). Replacements were found through referral; women who were referred were screened for reading comprehension skills, motivation and availability. If they met established criteria, they were offered the opportunity to become home visitors.

The intervention staff included a local field coordinator, six mentors, and a team of home visitors. The field coordinator oversaw the roll out of the intervention and monitored the mentors from the central office (in Bogota) and also on site. Six mentors, with an undergraduate degree in psychology/social work or comparable fieldwork experience, were recruited to train and supervise home visitors throughout the study. They had six weeks' pre-service training focused on the home visiting curriculum and protocols, training and supervision skills, creating home-made toys, and supervised practice. Four short (oneto four-day) refresher and feedback sessions took place in Bogota during the course of the intervention. These also provided the mentors the opportunity to exchange experiences, challenges and solutions. In addition, the mentors were in regular email/phone communication with the field coordinator, and email communication with a member of the research team, as needed.

Each mentor trained and supervised 24 home visitors, covering eight municipalities. The home visitors' pre-service training on the stimulation component lasted two weeks, with an additional week of in-service training after the team of mentors had completed the first round of itinerant supervision (one to two months after the intervention roll-out). The home visitor training sessions included: (1) basic concepts of child development and early learning; (2) the aims of the stimulation intervention and the role of the home visitor in this; (3) an introduction to play activities, the steps to demonstrate and scaffold each activity and guidelines to assess the progress of the child towards the learning goals; (4) guidelines to help manage difficult children; (5) a toy making workshop; (6) guidelines for record keeping. The training of home visitors emphasized the importance of a good working relationship with the beneficiary mother, positive reinforcement, and listening skills.

In addition, home visitors participated in a 5-hour training session on the micronutrient component of the intervention before rollout. The training sessions included a description of the micronutrient supplement, storage requirements, instructions for use, potential side effects, toxicity risks, and safety protocols in case of side effects.

# A.3 Rollout and Monitoring

The intervention was rolled out over four months from February through May 2010, and phased out 18 months later, from September through December 2011. Training and supervision was rolled out by geographical location, evenly across treatment groups, following baseline data collection. Once the intervention was up and running, mentors visited intervention communities once every 7 to 10 weeks to monitor implementation, provide support, and reinforce home visitors' motivation. At this time, mentors also distributed one-page bulletins to home visitors, with reminders of best practices in home visiting. In addi-

tion, mentors sent short text (SMS) messages to home visitors every month to reinforce key aspects the stimulation protocol. Home visitors were encouraged to call mentors for advice where necessary (calling cards were provided by the study team).

To monitor the psychosocial stimulation component, home visitors filled in a form at the end of each visit with basic information on the visit (e.g. date, activities performed, who present), as well as a short assessment of children's performance and engagement with the activities. Home visitors were paid \$100,000 Colombian pesos (COP) (19.4% of legal monthly minimum wage for 2010) per month.

To monitor the micronutrient component, the home visitor collected the empty sachets and intake charts (monitoring forms) from the household every two weeks. During these visits, mentors checked that tracking charts were filled in correctly, reminded the mother about the protocols of use, how to react if side effects arose and addressed any other queries and concerns. The home visitor was paid \$25,000 Colombian pesos (COP) per month for these activities.

# **B** Measures of skills and investments

In this section, we provide detailed information on each of the instruments we used to measure children's cognitive and socio-emotional skills, mother's cognitive and socio-emotional skills, and parental investments. In subsection B.4, we describe the non-parametric procedure we followed to standardize the measures of child and mother's skills for age.

# B.1 Measures on the target child

# B.1.1 Bayley Scales of Infant and Toddler Development, third edition (Bayley-III) (Bayley, 2006)

We administered the cognitive, receptive language, expressive language, fine motor and gross motor scales of the Bayley-III both at baseline and follow-up, following standard procedures. The scales assess children from birth up to 42 months by direct observation of performance on a series of items and are considered by many the "gold standard" for the assessment of children of these ages (Fernald et al., 2009). Bayley-III subscales were translated into Spanish, back translated to English to ensure accuracy, and piloted by testers. Children were assessed in local community centers with their mothers present. Testers held degrees in psychology and had a six-week training, including practice sessions with children of the target age groups. Inter-rater reliability (intra-class correlation) was above 0.9 on each subscale. Furthermore, 5% of the measurements were supervised by the trainer (reliabilities above 0.9) and corrective feedback was given when appropriate.

# B.1.2 MacArthur-Bates Communicative Development Inventories I, II and III - Spanish short Forms (SFI, SFII, SFIII)

We assessed language comprehension and production using the short-form versions of the Spanish Communicative Development Inventories. This is a parent report inventory and was collected in the house as part of the household survey. At baseline, we administered Spanish short forms of Inventories I and II to children of 12-18 and 19-24 months of age, respectively, which have been validated in Mexico (Jackson-Maldonado et al., 2012, 2003). At follow-up, we administered the Spanish short form of Inventory III to children 30-42 months. This form was under validation at the time of data collection (Jackson-Maldonado, 2011; Jackson-Maldonado and Conboy, 2011). We collaborated with the developer of the test in Spanish in the identification of suitable words in Colombian Spanish, prior and during piloting activities. We administered the vocabulary checklist (words the child "understands" and words the child says) for all Short Forms (SFs) and sentence structure sections (for SFIII only), and counted the number of words the child could say (as reported by the mother/caregiver) and number of more complex sentence structures the child uses.

# B.1.3 Infant Characteristics Questionnaire (ICQ) (Bates et al., 1979)

The Infant Characteristics Questionnaire (ICQ) assesses the construct of "difficult" temperament by maternal (caregiver) report. As such it measures parents' perceptions of the infant, not necessarily the infant's behavior as it might be objectively recorded. Both at baseline and follow-up, we used those items (17 in total) in questionnaires for 13-months and 24-months old children that related to the following constructs: difficult, unadaptable, unstoppable, and unsociable, as part of the household questionnaire. We made minor adjustments to the Spanish translations of the forms in order to maximise comprehension and the test predictive ability. For simplicity, we converted the 7-point rating items into 5-point ratings. We discussed these modifications with the author over email correspondence and piloted them before use in the field.

# B.1.4 Early Childrens Behavior Questionnaire (ECBQ) (Putnam et al., 2006, 2010)

At follow up, we complemented the assessment of temperament with measures of attention and inhibitory control by maternal report using the attentional focusing (sustained duration of orienting on an object of attention; resisting distraction), attentional shifting (the ability to transfer attentional focus from one activity/task to another) and inhibitory control (the capacity to stop, moderate, or refrain from a behavior under instruction) sub-scales in the short versions of the Spanish translation of the ECBQ. The ECBQ is designed to measure temperament in children aged 3-7 years. As before, for simplicity, we converted the 7-point rating scale into a 5-point rating scale. Minor language modifications to wording and sentence structure, with the aim to better reflect Colombian Spanish, were extensively piloted in the field.

### **B.2** Measures on the mother

### B.2.1 Maternal vocabulary

We assessed maternal receptive vocabulary in the first follow-up survey. For this, we used a selection of 50 words from the Spanish version of the Peabody Picture Vocabulary Test (PPVT), the Test de Vocabulario en Imagenes de Peabody (TVIP) (Dunnn et al., 1986). The reason why we used a selection of the words is because the Spanish version of the test is developed for ages 2.5-18 years. Hence, we could not use the test as designed (using established start and stopping rules) on our sample of mothers. Instead, we selected those words exhibiting a reasonable level of varying difficulty, after extensive piloting, and administered them all in the order in which they appear in the test. For each word, the subject points at the one picture (out of four) that best relates to the word (noun, action, abstract concept, adjective) that the tester calls out. The test was administered in the home by the interviewer at the end of the household interview

# B.2.2 Standard Progressive Matrices (RPM) (Raven, 1981)

We used the RPM system to measure mothers' reasoning ability or what is often referred as general intelligence in the second follow-up. This is a nonverbal test typically made of multiple choice items listed in order of difficulty, requiring ever greater capacity to encode and analyze the information, and which are organised in sets. In each test item, the subject is asked to identify the missing element that completes a pattern. Test items are presented in black ink on a white background. Out of the 5 sets of 12 matrices, we administered the middle 3 – hence, those with an intermediate level of difficulty. This decision was made after careful piloting.

We collected RPM at second follow up only under the presumption that maternal reasoning ability should not be affected by the intervention, and because of the monetary and time restrictions we faced in previous rounds.

# B.2.3 Center for Epidemiological Studies Depression scale (CES-D) (Andresen et al., 1994; Gonzalez et al., 1995; Radloff, 1977)

We assessed maternal depressive symptoms by direct interview with the mother using the 10-item Spanish version of the CES-D, the CESD-10, by Radloff (1977). We used the same measure both at baseline and follow up.

# B.3 Measures of parental investments: Family Care Indicators (FCI) (Frongillo et al., 2003)

The quality of the home environment (or level of stimulation in the home) was measured using items in the Family Care Indicators (FCI) developed by UNICEF (Frongillo et al., 2003). The FCI has been validated against the Home Observations for Measurement of the Home Environment (HOME) (Caldwell and Bradley, 2001), against which it was validated in Bangladesh (Hamadani et al., 2010).

Both at baseline and follow up, we collected by direct observation during the household survey the following information: the number of books for adults, the number of newspapers and magazines, and the number of varieties of play materials in the home that the child often played with. We collected by maternal (caregiver) report the number of play activities the child engaged in with an adult over the three days before the interview. Play materials include toys that make/play music; toys/objects meant for stacking, constructing or building; things for drawing, writing, colouring, and painting; toys for moving around; toys to play pretend games; picture and drawing books for children; and toys for learning shapes and colours. Play activities include reading or looking at picture books; telling stories to child; singing songs with child; playing with child with her toys; spending time with child scribbling, drawing, or colouring; and spending time with child naming things or counting; and taking child out for a leisure walk.

### **B.4** Age standardization of the measures

Total raw scores are increasing in age. Since we are interested in within sample comparisons, we internally standardize scores to remove the effect of age (child's age for the childs' measures and mother's age for the mothers' measures). We compute internal z-scores using the empirical age-conditional means and standard deviations estimated using non-parametric regression methods. In particular, for each measure to standardize, we compute the ageconditional mean using the fitted values of the following regression, estimated by kernel-weighted local polynomial smoothing methods:

$$Y_i = f(X_i) + \epsilon_i \tag{1}$$

where  $Y_i$  is the raw score of individual *i* in a given scale and  $X_i$  is the age of the individual (in months for the child, in years for the mother). Next, we regress the square of the residuals in equation (1) on age of the child as shown in the following kernel-weighted local polynomial regression:

$$(Y_i - \hat{f}_i)^2 = g(X_i) + v_i$$
(2)

Our estimate of the age-conditional standard deviation is the square root of the fitted values in equation (2). Finally, we compute the internally ageadjusted z-score,  $ZY_i$ , by subtracting from the raw score the within sample age-conditional mean estimated in (1) and dividing by the within sample ageconditional standard deviation obtained from (2). More specifically:

$$ZY_i = \frac{(Y_i - \hat{f}_i)}{\sqrt{\hat{g}_i}} \tag{3}$$

This procedure, less sensitive to outliers and small sample sizes within age category, resulted in smooth normally distributed internally standardized scores, with mean zero across the age range.

# C Exploratory factor analysis

This appendix describes the exploratory factor analysis (EFA) we performed to inform the specification of the measurement system described in Table 2 of the paper. Following the psychometric literature (Gorusch, 1983, 2003), we aim to build a measurement system with dedicated measures (measures that only proxy one latent factor) as it makes interpretation of the latent factors easiest and most transparent. EFA consists of two main steps. First, we select the number of latent factors that should be extracted from all the measures we have on each of the aspect we want to measure (e.g. child's development at baseline, child's development at follow-up, etc.). Second, we allocate measures to factors, estimate factor loadings and discard measurements that load on multiple factors in order to achieve a dedicated measurement system.

# C.1 Selecting the number of latent factors

The first step aims to determine how many latent factors should be extracted from each set of measures we have available to measure the child's development at baseline, the child's development at follow-up, parental investment at followup, the mother's skills, and the household's wealth at baseline. A variety of methods are available to select the number of factors, and here we implement four of the most popular methods developed in the literature. Below, we succinctly describe each one of them, before commenting on the number of factors they suggest to extract.

# C.1.1 Description of methods

**Kaiser's eigenvalue rule** The Kaiser's criterion consists in retaining only factors with eigenvalues greater than 1 (Kaiser, 1960). The intuition behind this rule is that unless a factor extracts at least as much variance as the equivalent of one original variable, it should be dropped.

**Cattell's scree plot** The scree test was proposed by Cattell (1966) and is based on the analyst's inspection of a plot of the eigenvalues associated with the data. Cattell's rule is such that the number of factors should be equal to the number of eigenvalues before which the smooth decrease of eigenvalues appears to level off to the right of the plot.

Velicer's minimum average partial (MAP) correlation rule Velicer (1976)'s minimum average partial (MAP) involves a complete factor analysis followed by the examination of a series of matrices of partial correlations. In the first step, the first factor is partialed out of the correlations between the variables of interest, and the average squared coefficient in the off-diagonals of the resulting partial correlation matrix is computed. In the second step, the first two factors are partialed out of the original correlation matrix and the average squared partial correlation is again computed. These computations are conducted for k-1, where k is the number of measurements. The number of components is determined by the step number in the analyses that resulted in the lowest average squared partial correlation. Intuitively, components are retained as long as the variance in the correlation matrix represents systematic

variance. Components are no longer retained when there is proportionately more unsystematic variance than systematic variance.

Horn's parallel analysis Horn's parallel analysis, described in Horn (1965), involves extracting eigenvalues from random data sets that parallel the actual data set with regard to the number of observations and variables. For example, if the original data set consists of n observations for each one of m variables, then a series of random data matrices of size  $n \times m$  are generated, and eigenvalues are computed for the correlation matrices for the original data and for each of the random data sets. The eigenvalues derived from the actual data are then compared to the eigenvalues derived from the random data. Factors are retained as long as the *i*-th eigenvalue from the actual data is greater than the *i*-th eigenvalue from the random data.

# C.1.2 Results

Appendix Table 1 reports how many factors each method suggests we should extract from all the measures we have available to measure child's development at follow-up and at baseline, parental investments, mother's skills and household's wealth. Most methods indicate that two factors should be extracted from the measures of child's development at follow-up. Between 1 and 3 factors should be extracted from the measures of child's development at baseline. Most methods also suggest that two factors should be extracted from the measures of parental investments and from the measures of maternal skills. Finally, only one factor seem underlie the measures of household wealth, according to all methods.

Table A1: Exploratory factor analysis to determining the number of factors

	Number of factors according to the following methods:			
	Kaiser's	Cattell's scree	Velicer's MAP	Horn's parallel
Dimensions to measure:	eigenvalue rule	plot	rule	analysis
Child's skills at $t+1$	2	2	2	3
Child's skills at t	1	2	1	3
Parental investments at $t+1$	2	2	2	3
Mother's skills	2	2	2	4
Wealth	1	1	1	3

The results from this first step of the EFA suggests that the data we work with may be rich enough to support the model we set out in Section 3, which assumes two dimensions for the child's skills, two dimensions for the mother's skills, and two dimensions for parental investments. We now need to estimate factor loadings to allocate groups of measures to different factors and identify measures that do not proxy one factor in order to finalise the configuration of measurement system and interpret each factor with precision. This is what we do in the second step of the EFA, which we describe now.

### C.2 Specifying the dedicated measurement system

Once we have a strong indication how many factors should be retained from the data, we search for dedicated measures for each factor by implementing an exploratory factor analysis with *quartimin* rotation. We first estimate the factor loadings in a measurement system for each of the elements we want to measure. We then rotate the factor loadings so as to identify measures that heavily load on one factor and are therefore good candidates for the dedicated measurement system.<sup>1</sup>

The aim of the *quartimin* rotation is to re-weight the factor loadings obtained from the EFA in a way that leads to a structure of factor loadings such that measures only heavily load on one factor. This helps in identifying good candidate measures for our system of dedicated measures. In contrast, if a measure does not load heavily on a factor or if it is not clearly related to only one factor, it cannot serve as a dedicated measure. In this case, we exclude it from our measurement system.

Table A2 reports rotated factor loadings for each measure. Note that we have assumed two factors for the measures of child development at baseline and at follow-up, two factors for the measures of mother's skills, two factors for the measures of parental investments and one factor for the measures of household wealth. Below we comment on these results and how they informed the final configuration of our measurement system.

Measures of child's development at t and t+1 The factor loadings on the measures of child's skills at follow-up (t+1) clearly suggest two groupings of measures. The Bayley measures and the Mac Arthur measures heavily load on a first factor, which we call cognitive skill. Some of the Bates measures and some of the Rothbart measures heavily loads on a second factor, which we call socio-emotional skill. Note that the measures "Bates unadaptable"

<sup>&</sup>lt;sup>1</sup>Several methods are available to rotate the factors. We focus on the results of a *quartimin* rotation because it is an oblique rotation and hence allows factors to be correlated. We also performed the same exercise using *geomin* rotation, since it is another type of oblique rotation, and reached similar conclusions with respect to the final measurement system.
and "Rothbart Sociable" load slightly more heavily on the first factor than on the second, but overall have two very small loadings of each of these factors (below 0.3) and do not clearly load on one of the two factors. As a result, we discard these two measures from the final measurement system. Based on these groupings, we interpret the first factor as measuring the child's cognitive skill at follow-up and the second factor as measuring the child's socio-emotional skill at follow-up.

Turning to the measures on the child's skills at baseline (t), we again obtain a pattern of factor loadings that clearly support two groupings of measures. On the one hand, all the Bayley measures (cognitive, language reception, language expression and motor fine) heavily load on a first factor, which we interpret as measuring the child's cognitive skill at baseline. The Mac Arthur-Bates measures of number of words understood has two very small loadings and does not clearly load on one of the two factors. As a result, we discard this measure as an unfit candidate for our dedicated measurement system. The Mac Arthur-Bates measures of number of spoken shows a s clearer pattern, with a heavier loading on factor 1 than factor 2. Although this pattern is not as clear as the Bayley measures, we do retain Mac Arthur-Bates measure of the number of words spoken by the child in our final measurement system, so as to have a measure of the child's vocabulary measuring cognitive skill both at baseline and at follow-up. Finally, the Bates measures of the child's temperament at baseline clearly load on the second factor. In the final measurement system, we retain these four measures to proxy the factor that we call socio-emotional skill of the child at baseline.

Measures of parental investments The measures of parental investments from the follow-up survey indicate two clear groupings. On the one hand, the variables measuring the variety of play materials and the number of different types of play material, for the most part, load on a first factor. On the other hand, the variable measuring the variety of play activities and most of the variables measuring the frequency of each of these activities in the three days preceding the interview load on a second factor. A few of the variables do not clearly load on one the factors, and for this reason, we exclude from our system of dedicated measures. In particular, we exclude "Number of picture books", "Number of home-made toys", and "Times the mother went outside with the child in the last 3 days" from our final measurement system.

Measure of maternal skills The measures of maternal skills that we have at our disposal in the survey indicate two clear groupings of measure. On the one hand, the mother's years of education, vocabulary and IQ score, along with two items from the FCI (the number of books in the house and the number of magazines and newspapers) seem to be highly correlated to each other, as they clearly load on a first factor (Factor 2 in the table). On the other hand, most items of the CES-D scale heavily load on a second factor. The only exception is the third measure asking the mother "Did you feel hopeful about the future?", and we exclude this measure from our final measurement system since it is not clearly dedicated to one of the two factors measuring maternal skills. Measure of household wealth The baseline survey contains a multitude of measures that could provide information about the household's underlying wealth level. This includes measures of the household's assets (dwelling, car, cellphone, etc.), along with characteristics of the household's dwelling (dirt floors, solid walls) and a measure of the dwelling crowding. The rotated factor loadings for all these measures indicate that most relate strongly to the underlying factor. However, there are a few exceptions and we rid the final measurement system of the measures that have a rotated factor loading below 0.25.

Table A2:	Exploratory	factor	analysis -	Estimates	of rotated	factor	loadings
100010 1120		1000001	~~~~~ J D1D		51 100000a	100001	

	Factor 1	Factor 2
Bayley Cognitive	0.755	0.021
Bayley Receptive Language	0.730	0.055
Bayley Expressive Language	0.752	0.039
Bayley Fine Motor	0.611	0.036
Mac Arthur-Bates Vocabulary	0.628	0.169
Mac Arthur-Bates Complex Sentences	0.529	0.176
Bates Unsociable sub-scale (-)	0.233	0.299
Bates Difficult sub-scale (-)	0.165	0.662
Bates Unadaptable sub-scale (-)	0.224	0.179
Bates Unstoppable sub-scale (-)	0.074	0.662
Rothbart Inhibitory Control	0.154	0.678
Rothbart Attention	0.124	0.340
Rothbart Sociable	0.236	0.117

Rotated factor loadings on measures of ch	ild's skills a	t <i>t</i>
	Factor 1	Factor 2
Bayley Cognitive	0.680	0.080
Bayley Receptive Language	0.739	-0.015
Bayley Expressive Language	0.723	0.048
Bayley Fine Motor	0.716	-0.041
Mac Arthur-Bates Vocabulary	0.243	0.148
Mac Arthur-Bates Comprehension	0.049	0.117
Bates Difficult sub-scale (-)	0.113	0.533
Bates Unadaptable sub-scale (-)	0.157	0.413
Bates Unstoppable sub-scale (-)	-0.153	0.393
Bates Unsociable sub-scale (-)	0.212	0.344

Rotated factor loadings on measures of paren	tal	investi	nents
			_

	Factor 1	Factor 2
Number of different play materials	0.830	0.271
Number of picture books	0.336	0.166
Number of paint books	0.420	0.120
Number of home-made toys	0.140	0.117
Number of toys bought	0.582	0.108
Number of toys to move	0.607	0.160
Number of toys to learn shapes	0.571	0.184
Number of different play activities in the last 3 days	0.299	0.828
Times told a story to child in last 3 days	0.124	0.538
Times read to child in last 3 days	0.144	0.546
Times went outside with the child in the last 3 days	0.170	0.310
Time played with toys and child in last 3 days	0.200	0.522
Time named things to child in last 3 days	0.154	0.525

Table A2: Exploratory factor analysis - Estimates of rotated factor loadings (continued)

Rotated factor loadings on measures of mother's	skills	
	Factor 1	Factor 2
Mothers' years of education*	0.042	0.666
Mother's vocabulary	0.106	0.707
Number of books in the house	0.011	0.456
Number of magazines and newspapers	-0.012	0.305
Raven's score ("IQ") **	0.075	0.661
Were you bothered by things that usually don't bother you?	0.509	0.016
Had you trouble keeping your mind on what you were doing?	0.493	0.078
Did you feel hopeful about the future	0.074	0.207
Did you feel depressed?	0.611	0.159
Did you feel everything you did was an effort?	0.513	0.118
Did you feel fearful?	0.510	-0.027
Was your sleep restless?	0.514	-0.047
Did you feel happy?	0.360	0.164
How often did you feel lonely in the last 7 days?	0.527	0.027
Did you feel you couldn't get going?	0.610	0.051

Rotated factor loadi	gs on measures of	f household	wealth at t
----------------------	-------------------	-------------	-------------

	Factor 1
Owns dwelling?	0.299
Dwelling has dirt floors?	-0.219
Dwelling has solid walls?	0.200
Number of people in the dwelling	0.290
Dwelling has sewage system?	0.191
Owns a car?	0.303
Owns a cellphone?	0.349
Owns a computer?	0.265
Owns a fridge?	0.601
Owns a washing machine?	0.409
Owns a blender?	0.495
Owns a TV?	0.392
Owns a radio?	0.338
Owns a fan?	0.172

## C.3 Estimates of the measurement system

The following tables report the estimates of the measurement system. Appendix Table 3 reports the estimates of the factor loadings in each measurement equation. Appendix Table 4 reports the estimates of the means of the latent factor distributions for the treated households relative to the means of the latent factor distributions for the control households (which is normalized to 0). Appendix Table 5 reports the estimates of the variance of and correlation between the latent factors for the treated and control households separately

	en
	syst
-	measurement
	the
•	ID
-	loadings
	lactor
د	01
- F	Estimates
	A3:
-	able
F	Ĩ

					Latent factors				
	Child's cognitive wills (++1)	Child's cognitive Skill (1)	Child's socio- emotional skill	Child's socio- emotional skill	Material investment	Time investment	Mother's connitive skill	Mother's socio- emotional skill	Wealth (t)
Measures:			(t+I)	(t)			0		
Boular Comitive	1 000	0	0	C	0	0	C	0	C
	1.000	0 0		0 0	0 0		0 0		
Bayley Receptive Language	076.0	0	0	0	0	0	0	0	0
Bayley Expressive Language	1.048	0	0	0	0	0	0	0	0
Bayley Fine Motor	0.768	0	0	0	0	0	0	0	0
Mac Arthur-Bates Vocabulary	0.735	0	0	0	0	0	0	0	0
Mac Arthur-Bates Complex Sentences	0.569	0	0	0	0	0	0	0	0
Bayley Cognitive	0	1.000	0	0	0	0	0	0	0
Bayley Receptive Language	0	1.031	0	0	0	0	0	0	0
Bayley Expressive Language	0	1.056	0	0	0	0	0	0	0
Bayley Fine Motor	0	0.953	0	0	0	0	0	0	0
Mac Arthur-Bates Vocabulary	0	0.576	0	0	0	0	0	0	0
Bates Difficult sub-scale (-)	0	0	1.000	0	0	0	0	0	0
Bates Unsociable sub-scale (-)	0	0	0 444	0	0	0	0	- 0	C
Bates IInstannahle sub-scale (_)			0.948	~ ~					
Dathhart Inhihitany Control cub coale			1 007	~					
Dothhort Attention cub could			0.505						
	0 0	0		000,	0 0	0	0 0	0 0	0 0
Bates Difficult factor (-)	0	0	0	1.000	0	0	0	0	0
Bates Unsociable factor (-)	0	0	0	0.439	0	0	0	0	0
Bates Unadaptable (-)	0	0	0	0.639	0	0	0	0	0
Bates Unstoppable (-)	0	0	0	0.511	0	0	0	0	0
Number of different play materials	0	0	0	0	1.000	0	0	0	0
Number of colouring books	0	0	0	0	0.368	0	0	0	0
Number of toys bought	0	0	0	0	0.582	0	0	0	0
Number of toys that require movement	0	0	0	0	0.651	0	0	0	0
Number of toys to learn shapes	0	0	0	0	0.749	0	0	0	0
Number of different play activities	0	0	0	0	0	1.000	0	0	0
Times told a story to child in last 3 days	0	0	0	0	0	0.627	0	0	0
Times read to child in last 3 days	0	0	0	0	0	0.707	0	0	0
Times played with child and toys in last 3 days	0	0	0	0	0	0.597	0	0	0
Times labelled things to child in last 3 days	0	0	0	0	0	0.606	0	0	0
Mothers' vears of education	0	ò	° O	ò	) c	0	1 000	Ô	ò
Mother's years of curvation Mother's viscolaridary							1 040		
MULLES VOCADULALY Mumbra of hools for odulta in the house							1.040		
	0 0	0 0		0 0	0 0		101.0		
Number of magazines and newspapers					0 0		0.401		
Kaven's score (IQ)	0 0	0 0	0 0	0 0	0 0	0 0	0.090	0	0 0
Did you feel depressed? (-)	0 0	0 0	0	0 0	0 0	0 0	0 0	1.000	0 0
Bothered by what usually don't bother you? (-)	0	0 0	0 0	0 0	0 0	0 0	0	0.842	0 0
Had trouble keeping mind on doing? (-)	0	0 0	0	0 0	0	0	0	0.917	0 0
Felt everything you did was an effort? (-)	0	0	0	0	0	0	0	0.823	0
Did you teel fearful? (-)	0	0	0	0	0	0	0	0.695	0
Did you sleep was restless? (-)	0	0	0	0	0	0	0	0.798	0
Did you feel happy? (-)	0	0	0	0	0	0	0	0.497	0
How often did you feel lonely last week? (-)	0	0	0	0	0	0	0	0.825	0
Did you feel you couldn't get going? (-)	0	0	0	0	0	0	0	0.961	0
Owns a fridge	0	0	0	0	0	0	0	0	1
Owns a car	0	0	0	0	0	0	0	0	0.378
Owns a computer	0	0	0	0	0	0	0	0	0.945
Owns a blender	0	0	0	0	0	0	0	0	0.836
Own a washing maching	0	0	0	0	0	0	0	0	0.420
Owns dwelling	0	0	0	0	0	0	0	0	0.515
Owns a radio	0	0	0	0	0	0	0	0	0.535
Ows a TV	0	0	0	0	0	0	0	0	0.919

Table A4: Estimates of the difference in means of the latent factors between treated and controls

Latent factor	Mean	Standard	90% Confidence
		deviation	Internal
Child's cognitive skills at $t+1$	0.108	(0.055)	[0.015,0.186]
Child's cognitive skills at <i>t</i>	-0.031	(0.074)	[-0.159,0.084]
Child's socio-emotional skills at $t+1$	0.082	(0.044)	[0.008,0.149]
Child's socio-emotional skills at t	0.037	(0.068)	[-0.077,0.148]
Material investments at $t+1$	0.200	(0.072)	[0.072,0.303]
Time investments at $t+1$	0.345	(0.071)	[0.221,0.449]
Mother's cognitive skills	-0.026	(0.046)	[-0.104,0.047]
Mother's socio-emotional skills	0.007	(0.038)	[-0.063,0.063]

	Child's	Child's	Child's socio-	Child's socio-	Material	Time	Mother's	Mother's	Wealth (t)
	cognitive skills (t+1)	cognitive Skill (t)	emotional skill (t+1)	emotional skill (t)	investment	investment	cognitive skill	socio- emotional	
					Control group			SKIII	
Child's cognitive skills at t+1	1.00	0.64	0.35	0.30	0.40	0.28	0.50	0.13	0.24
Child's cognitive skills at t	0.64	1.00	0.27	0.26	0.28	0.24	0.35	0.08	0.17
Child's socio-emotional skills at t+1	0.35	0.27	1.00	0.62	0.36	0.28	0.29	0.14	0.12
Child's socio-emotional skills at t	0.30	0.26	0.62	1.00	0.32	0.26	0.44	0.16	0.24
Material investments at t+1	0.40	0.28	0.36	0.32	1.00	0.62	0.61	0.07	0.32
Time investments at t+1	0.28	0.24	0.28	0.26	0.62	1.00	0.41	-0.02	0.18
Mother's cognitive skills	0.50	0.35	0.29	0.44	0.61	0.41	1.00	0.19	0.44
Mother's socio-emotional skills	0.13	0.08	0.14	0.16	0.07	-0.02	0.19	1.00	0.02
Wealth at <i>t</i>	0.24	0.17	0.12	0.24	0.32	0.18	0.44	0.02	1.00
					<b>Freated</b> group				
Child's cognitive skills at t+1	1.00	0.68	0.32	0.21	0.30	0.17	0.27	0.12	0.14
Child's cognitive skills at t	0.68	1.00	0.26	0.22	0.24	0.14	0.21	0.08	0.09
Child's socio-emotional skills at t+1	0.32	0.26	1.00	0.43	0.29	0.28	0.24	0.13	0.08
Child's socio-emotional skills at t	0.21	0.22	0.43	1.00	0.10	0.05	0.29	0.15	0.24
Material investments at t+1	0.30	0.24	0.29	0.10	1.00	0.45	0.48	0.03	0.26
Time investments at t+1	0.17	0.14	0.28	0.05	0.45	1.00	0.29	-0.14	0.02
Mother's cognitive skills	0.27	0.21	0.24	0.29	0.48	0.29	1.00	0.12	0.40
Mother's socio-emotional skills	0.12	0.08	0.13	0.15	0.03	-0.14	0.12	1.00	0.18
Wealth at t	0.14	0.09	0.08	0.24	0.26	0.02	0.40	0.18	1.00

Table A5: Estimates of the correlation matrix of the latent factors for treated and controls

## D Specification and robustness checks

In this section of the appendix, we report results associated with specification and robustness checks we discuss in the main text.

Tables A6 and A7 report the estimates of the investment function and production functions when the investment function is fully interacted with the treatment indicator.

Tables A8 and A9 report the estimates of the CES production function for cognitive skills and socio-emotional skills, respectively, where all the share parameters (as well as the total productivity factor A) is interacted with the treatment.

Tables A10 and A11 report the estimates of the CES production function for cognitive skills and socio-emotional skills, respectively where the child's age at follow-up is allowed to determine investments (in the investment functions) and to affect total-factor productivity.

	Log of materi	ial investment	Log of time	investment
	Main effect	Interaction with Treatment	Main effect	Interaction with Treatment
Constant	-0.006	-0.001	0.257	0.343
	(0.020)	(0.027)	(0.082)	(0.075)
	[-0.03, 0.035]	[-0.033, 0.053]	[0.113, 0.389]	[0.202, 0.445]
Log of child's cognitive skills at t	0.083	0.124	0.089	-0.015
	(0.084)	(0.061)	(0.122)	(0.106)
	[-0.041, 0.236]	[0.015, 0.216]	[-0.13, 0.249]	[-0.186, 0.16]
Log of child's socio-emotional skills at t	0.068	0.129	-0.158	-0.163
	(0.095)	(0.100)	(0.127)	(0.139)
	[-0.063, 0.245]	[0.001, 0.323]	[-0.394, 0.019]	[-0.443, 0.012]
Log of mother's cognitive skills	0.753	0.364	-0.145	0.076
	(0.114)	(0.124)	(0.172)	(0.163)
	[0.563, 0.92]	[0.155, 0.567]	[-0.386,0.165]	[-0.193, 0.346]
Log of mother's socio-emotional skills	-0.092	-0.216	-0.089	-0.130
	(0.123)	(0.116)	(0.156)	(0.157)
	[-0.343, 0.053]	[-0.419,-0.034]	[-0.247,0.246]	[-0.366,0.156]
Log of wealth at <i>t</i>	0.043	-0.053	0.116	-0.032
	(0.103)	(0.147)	(0.147)	(0.186)
	[-0.135, 0.208]	[-0.343, 0.15]	[-0.097, 0.373]	[-0.266,0.337]
Mother is married at $t+I$	0.113	0.082	0.015	0.057
	(0.037)	(0.031)	(0.054)	(0.053)
	[0.053, 0.174]	[0.027, 0.132]	[-0.075, 0.105]	[-0.028, 0.147]
Log of number of children at $t+I$	-0.058	-0.143	-0.062	0.074
	(0.046)	(0.046)	(0.073)	(0.069)
	[-0.148,0]	[-0.221,-0.071]	[-0.16, 0.078]	[-0.024, 0.199]
Log of average male wages in village at $t+I$	0.007	0.087	0.084	-0.191
	(0.073)	(0.077)	(0.097)	(0.100)
	[-0.131, 0.105]	[-0.038, 0.209]	[-0.071,0.245]	[-0.368,-0.047]
Log of average female wages in village at $t+I$	0.014	-0.010	-0.018	0.079
	(0.066)	(0.072)	(0.084)	(0.091)
	[-0.079, 0.136]	[-0.145, 0.094]	[-0.191, 0.091]	[-0.046,0.245]

Notes: Standard errors in parentheses and 90% confidence intervals in brackets based on 1000 bootstraps.

Table A6: Estimates of the fully interacted log-linear investment function

	Cognitive	Socio-emotional
	skills	skills
Child's cognitive skills at t	0.567	0.104
	(0.051)	(0.054)
	[0.499,0.667]	[0.012,0.189]
Child's socio-emotional skills at t	0.037	0.416
	(0.046)	(0.057)
	[-0.035,0.118]	[0.355,0.532]
Mother's cognitive skills	0.053	0.017
	(0.104)	(0.120)
	[-0.122,0.235]	[-0.254,0.136]
Mother's socio-emotional skills	0.052	0.155
	(0.047)	(0.057)
	[-0.023,0.123]	[0.043,0.229]
Material investments at $t+1$	0.358	-0.155
	(0.141)	(0.150)
	[0.1,0.551]	[-0.278,0.211]
Time investments at $t+1$	-0.114	0.384
	(0.107)	(0.108)
	[-0.292,0.061]	[0.158,0.511]
Number of children in household at $t+1$	0.047	0.078
	(0.029)	(0.027)
	[-0.002,0.092]	[0.028,0.119]
Control function for material investments	-0.298	0.317
	(0.160)	(0.161)
	[-0.532,-0.001]	[-0.051,0.459]
Control function for time investment	0.137	-0.3
	(0.126)	(0.118)
	[-0.07,0.333]	[-0.461,-0.079]
Complementarity parameter	0.08	0.011
	(0.065)	(0.058)
	[-0.031,0.179]	[-0.067,0.122]
Elasticity of substitution	1.087	1.012
	(0.077)	(0.065)
	[0.97,1.218]	[0.938,1.139]
Productivity parameter (A)	0.991	0.993
	(0.012)	(0.013)
	[0.968,1.007]	[0.974,1.015]
Productivity parameter interacted with treatment	0.08	-0.037
	(0.066)	(0.053)
	[-0.009,0.206]	[-0.13,0.041]

Table A7: Estimates of the CES production for cognitive and socio-emotional skills with a fully interacted log-linear investment function

	Without control	With control
	function	function
Child's cognitive skills at t	0.548	0.524
	(0.063)	(0.071)
	[0.455,0.665]	[0.435,0.665]
Child's cognitive skills at $t *$ Treat	0.072	0.064
	0.088	0.088
	[-0.068,0.221]	[-0.078,0.211]
Child's socio-emotional skills at t	0.02	0.033
	(0.064)	(0.071)
	[-0.078,0.14]	[-0.068,0.164]
Child's socio-emotional skills at t * Treat	0	-0.005
	0.084	0.085
	[-0.142,0.128]	[-0.147,0.128]
Mother's cognitive skills	0.281	0.125
-	(0.085)	(0.173)
	[0.16,0.429]	[-0.151,0.416]
Mother's cognitive skills * Treat	-0.172	-0.151
fiolder s cognitive skins - from	0.108	0.112
	[-0.371,-0.023]	[-0.351,0.011]
Mother's socio-emotional skills	0.042	0.042
	(0.079)	(0.076)
	[-0.111,0.142]	[-0.106,0.147]
Mother's socio-emotional skills * Treat	0.054	0.04
	0.095	0.094
	[-0.088,0.224]	[-0.103,0.21]
Material investments at $t+1$	0.082	0.375
	(0.063)	(0.236)
	[-0.023,0.19]	[0.039,0.782]
Material investments at $t+1$ *Treat	0.011	0.001
	0.078	0.08
	[-0.121,0.135]	[-0.137,0.125]
Fime investments at $t+1$	0	-0.141
	(0.062)	(0.153)
	[-0.12,0.082]	[-0.454,0.031]
Time investments at $t+1$ * Treat	0.015	0.029
	0.076	0.077
	[-0.098,0.152]	[-0.082,0.17]
Number of children in household at $t+1$	0.026	0.041
	(0.042)	(0.044)
	[-0.045,0.094]	[-0.033.0.112]
Number of children in household at $t+1$ * Treat	0.047	0.062
	0.033	0.036
	[-0.008.0.094]	[-0.002.0.116]

Table A8: Estimates of the CES production for cognitive skills where all the share parameters are interacted with the treatment

	Without control	With control
	function	function
Control function for material investments	-	-0.3
		0.229
		[-0.692,0.02]
Control function for time investment	-	0.14
		0.15
		[-0.048,0.434]
Complementarity parameter	0.143	0.057
	0.081	0.061
	[-0.035,0.229]	[-0.018,0.169]
Elasticity of substitution	1.166	1.061
	0.105	0.073
	[0.966,1.297]	[0.982,1.203]
Productivity parameter (A)	0.978	0.987
	0.012	0.011
	[0.966,1.005]	[0.97,1.007]
Productivity parameter interacted with treatment	0.097	0.08
	0.054	0.075
	[0.016,0.189]	[-0.018,0.224]

Table A8 : Estimates of the CES production for cognitive skills where all the share parameters are interacted with the treatment (continued)

	Without control	With control
	function	function
Child's cognitive skills at t	0.123	0.12
	(0.067)	(0.079)
	[0.016,0.238]	[-0.002,0.256]
Child's cognitive skills at $t *$ Treat	-0.013	-0.011
	0.084	0.084
	[-0.147,0.126]	[-0.147,0.134]
Child's socio-emotional skills at t	0.557	0.531
	(0.090)	(0.092)
	[0.473,0.761]	[0.453,0.746]
Child's socio-emotional skills at t * Treat	-0.247	-0.239
	0.096	0.097
	[-0.449,-0.125]	[-0.441,-0.121]
Mother's cognitive skills	-0.144	-0.032
	(0.105)	(0.191)
	[-0.33,0.013]	[-0.371,0.253]
Mother's cognitive skills * Treat	0.16	0.131
	0.13	0.136
	[-0.051,0.384]	[-0.087,0.362]
Mother's socio-emotional skills	0.168	0.16
	(0.086)	(0.084)
	[0.018,0.299]	[0.016,0.294]
Mother's socio-emotional skills * Treat	0.01	0.03
	0.104	0.105
	[-0.165,0.165]	[-0.151,0.175]
Material investments at $t+1$	0.196	-0.172
	(0.075)	(0.236)
	[0.074,0.316]	[-0.537,0.258]
Material investments at $t+1$ *Treat	-0.082	-0.077
	0.088	0.09
	[-0.216,0.068]	[-0.209,0.078]
Time investments at $t+1$	0.048	0.37
	(0.067)	(0.156)
	[-0.065,0.154]	[0.052,0.546]
Time investments at $t+1$ * Treat	0.091	0.084
	0.083	0.084
	[-0.033,0.241]	[-0.043,0.236]
Number of children in household at $t+1$	0.052	0.022
	(0.044)	(0.044)
	[-0.036,0.111]	[-0.052,0.091]
Number of children in household at $t+1$ * Treat	0.133	0.104
	0.033	0.035
	[0.082,0.19]	[0.059,0.17]

Table A9: Estimates of the CES production for socio-emotional skills where all the share parameters are interacted with the treatment

	Without control	With control
	function	function
Control function for material investments	-	0.379
		(0.223)
		[-0.038,0.724]
Control function for time investment	-	-0.34
		(0.145)
		[-0.496,-0.033]
Complementarity parameter	-0.023	-0.042
	(0.069)	(0.051)
	[-0.118,0.102]	[-0.101,0.06]
Elasticity of substitution	0.977	0.959
	(0.069)	(0.050)
	[0.895,1.114]	[0.908,1.064]
Productivity parameter (A)	1.017	1.021
	(0.012)	(0.009)
	[0.981,1.019]	[0.987,1.017]
Productivity parameter interacted with treatment	-0.027	-0.056
	0.039	0.057
	[-0.082,0.046]	[-0.118,0.061]

Table A9 (cont.): Estimates of the CES production for socio-emotional skills where all the share parameters are interacted with the treatment

	Without control	With control
	function	function
Child's cognitive skills at t	0.593	0.565
	(0.044)	(0.057)
	[0.524,0.67]	[0.485,0.677]
Child's socio-emotional skills at t	0.032	0.038
	(0.043)	(0.050)
	[-0.037,0.106]	[-0.033,0.125]
Mother's cognitive skills	0.179	0.018
	(0.049)	(0.127)
	[0.103,0.263]	[-0.176,0.231]
Mother's socio-emotional skills	0.066	0.055
	(0.045)	(0.049)
	[-0.011,0.131]	[-0.026,0.125]
Material investments at $t+1$	0.09	0.418
	(0.033)	(0.207)
	[0.034,0.145]	[0.095,0.744]
Time investments at $t+1$	0.006	-0.145
	(0.035)	(0.141)
	[-0.055,0.059]	[-0.392,0.047]
Number of children in household at $t+1$	0.034	0.051
	(0.026)	(0.030)
	[-0.009,0.078]	[0.001,0.098]
Control function for material investments	-	-0.344
		(0.217)
		[-0.673,-0.006]
Control function for time investment	-	0.16
		(0.150)
		[-0.053,0.421]
Complementarity parameter	0.149	0.056
	(0.079)	(0.061)
	[-0.031,0.229]	[-0.036,0.164]
Elasticity of substitution	1.175	1.059
	(0.101)	(0.071)
	[0.97,1.296]	[0.965,1.196]
Productivity parameter (A)	0.984	0.998
	(0.018)	(0.016)
~	[0.958,1.014]	[0.968,1.023]
Productivity parameter interacted with treatment	0.091	0.07
	0.053	0.073
	[0.029,0.201]	[-0.013,0.223]
Productivity parameter interacted with age	U (0.000)	-0.005
	(800.0)	(0.007)
	[-0.011,0.014]	[-0.013,0.009]

Table A10: Estimates of the CES production for cognitive skills, where age is allowed to affect total-factor productivity

	Without control	With control
	function	function
Child's cognitive skills at <i>t</i>	0.112	0.116
	(0.044)	(0.059)
	[0.04,0.188]	[0.025,0.218]
Child's socio-emotional skills at t	0.443	0.428
	(0.054)	(0.057)
	[0.368,0.551]	[0.349,0.538]
Mother's cognitive skills	-0.055	0.057
	(0.065)	(0.138)
	[-0.166,0.046]	[-0.169,0.281]
Mother's socio-emotional skills	0.146	0.155
	(0.057)	(0.057)
	[0.05,0.23]	[0.046,0.231]
Material investments at $t+1$	0.148	-0.22
	(0.042)	(0.192)
	[0.08,0.218]	[-0.526,0.091]
Time investments at $t+1$	0.112	0.398
	(0.041)	(0.131)
	[0.046,0.181]	[0.174,0.597]
Number of children in household at $t+1$	0.093	0.065
	(0.026)	(0.027)
	[0.049,0.135]	[0.027,0.115]
Control function for material investments	-	0.385
		(0.198)
		[0.061,0.703]
Control function for time investment	-	-0.31
		(0.135)
		[-0.505,-0.064]
Complementarity parameter	0.061	0.029
	(0.075)	(0.055)
	[-0.084,0.151]	[-0.049,0.116]
Elasticity of substitution	1.065	1.029
	(0.086)	(0.060)
	[0.923,1.178]	[0.953,1.131]
Productivity parameter (A)	0.992	0.996
	(0.020)	(0.018)
	[0.965,1.031]	[0.963,1.022]
Productivity parameter interacted with treatment	-0.027	-0.042
	0.041	0.05/
Deside statistics and a statistic statistic	[-0.081,0.056]	[-0.113,0.0/1]
Productivity parameter interacted with age	-0.002	0.004
	(0.007)	(0.008)
	[-0.013,0.009]	[-0.01,0.01/]

Table A11: Estimates of the CES production for socio-emotional skills, where age is allowed to affect total-factor productivity

## References

- Andresen, E., J. Malmgren, W. Carter, and D. Patrick (1994). Screening for depression in well older adults: evaluation of a short form of the ces-d (center for epidemiologic studies depression scale). American Journal of Preventive Medicine 10(758), 77–84.
- Attanasio, O., C. Fernandez, E. Fitzsimons, S. Grantham-McGregor, C. Meghir, and M. Rubio-Codina (2014). Using the infrastructure of a Conditional Cash Transfer programme to deliver a scalable integrated early child development programme in Colombia: A cluster randomised controlled trial. Forthcoming in *British Medical Journal 349*, g5785.
- Bates, J., C. Freeland, and M. Lounsbury (1979). Measurement of infant difficultness. *Child Development* 50, 794–803.
- Bayley, N. (2006). Bayley Scales of Infant and Toddler Development (3rd ed.). San Antonio, TX: Harcourt Assessment.
- Caldwell, B. and R. Bradley (2001). HOME Inventory and Administration Manual (Third Edition ed.). Little Rock, AK: University of Arkansas for Medical Sciences.
- Cattell, R. (1966). The scree test for the number of factors. *Multivariate* Behavioral Research 1, 629–637.
- Dunnn, L., E. Padilla, D. Lugo, and L. Dunn (1986). Manual del Examinador para el Test de Vocabulario en Imgenes Peabody (Peabody Picture Vocabulary Test), Adaptacin Hipanoamericana (Hispanic-American Adaption). Minneapolis, MS: Pearson Assessments.
- Fernald, L., P. Kariger, P. Engle, and A. Raikes (2009). Examining child development in low-income countries: A toolkit for the assessment of children in the first five years of life. Washington DC: The World Bank.
- Frongillo, E., S. Sywulka, and P. Kariger (2003). UNICEF psychosocial care indicators project. Final report to UNICEF. Mimeo, Cornell University.
- Gonzalez, V., A. Stewart, P. Ritter, and K. Lorig (1995). Translation and validation of arthritis outcome measures into Spanish. Arthritis Rheum 38, 1429–46.
- Gorusch, R. (1983). Factor Analysis. Hillsdale, NJ: Lawrence Erlbaum Associates.

- Gorusch, R. (2003). Factor analysis. In Handbook of psychology: Research methods in psychology, Volume 2, pp. 143–164. Hoboken, NJ: John Wiely & Sons, Inc.
- Hamadani, J., F. Tofail, A. Hilaly, S. Huda, P. Engel, and S. Grantham-McGregor (2010). Use of Family Care Indicators and their relationship with child development in bangladesh. *Journal of Health Population and Nutrition* 28(1), 23–33.
- Horn, J. (1965). A rationale and a test for the number of factors in factor analysis. *Psychometrika 30*, 179–185.
- Jackson-Maldonado, D. (2011). Spanish cdi-iii development and preliminary results. Presentation at the International Congress for the Study of Child Language, Montreal.
- Jackson-Maldonado, D. and B. Conboy (2011). Preliminary norms and validity for the Spanish CDI-III. American Speech-Language-Hearing Association (ASHA) Annual Conference. San Diego, CA.
- Jackson-Maldonado, D., V. Marchman, and L. Fernald (2012). Short form versions of the Spanish MacArthur-Bates Communicative Development Inventories. In Applied Psycholinguistics. Cambridge: Cambridge University Press.
- Jackson-Maldonado, D., D. Thal, V. Marchman, T. Newton, L. Fenson, and B. Conboy (2003). MacArthur Inventarios del Desarrollo de Habilidades Comunicativas: User's guide and technical manual. Baltimore, MD: Brookes.
- Kaiser, H. (1960). The application of electronic computers to factor analysis. Educational and Psychological Measurement 20, 141–151.
- Putnam, S. P., M. Gartstein, and M. K. Rothbart (2006). Measurement of fine-grained aspects of toddler temperament: The Early Childhood Behavior Questionnaire. *Infant Behavior and Development* 29(3), 386–401.
- Putnam, S. P., J. Jacobs, M. Garstein, and M. Rothbart (2010). Development and assessment of short and very short forms of the Early Childhood Behavior Questionnaire.
- Radloff, L. (1977). The CES-D scale: A self-report depression scale for research in the general population. Applied Psychological Measurement 1(3), 385– 401.

- Raven, J. (1981). Manual for Raven's progressive matrices and vocabulary scales. In Research Supplement No.1: The 1979 British Standardisation of the Standard Progressive Matrices and Mill Hill Vocabulary Scales, Together With Comparative Data From Earlier Studies in the UK, US, Canada, Germany and Ireland. San Antonio, TX: Harcourt Assessment.
- Velicer, W. (1976). Determining the number of components from the matrix of partial correlations. *Psychometrika* 41, 321–327.